This Settlement Agreement is dated the  $28^{\circ}$  day of September 2006 and is made between:

- (1) Mechanical-Copyright Protection Society Limited whose registered office is at 29-33.
  Berners Street, London W1T 3AB; and
- (2) Performing Right Society Limited whose registered office is at 29-33 Berners Street, London W1T 3AB; and
- (3) British Academy of Composers & Songwriters whose registered office is at 25-27 Berners Street, London W1T 3LR; and
- (4) The British Phonographic Industry Limited whose registered office is at Riverside Building, County Hall, Westminster Bridge Road, London SE1 7JA; and
- (5) iTunes S.à.r.I. whose registered office is at 8 rue Heinrich Heine, L-1720, Luxembourg; and
- (6) O2 (UK) Limited whose registered office is at 260 Bath Road, Slough, Berkshire, SL1 4DX; and
- (7) T-Mobile International (UK) Limited whose registered office is at Building 4, Hatfield Business Park, Hatfield, Hertfordshire AL10 9BW; and
- (8) Vodafone UK Content Services Limited whose registered office is at Vodafone House, The Connection, Newbury, Berkshire RG14 2FN; and
- (9) Orange Personal Communications Services Limited whose registered office is at St James Court, Great Park Road, Almondsbury Park, Bradley Stoke, Bristol, BS32 4QJ.

# Recitals

- A Terms used in these Recitals shall have the meanings ascribed to them in Clause 1 of this Settlement Agreement.
- B MCPS and PRS promulgated a series of interim, non-precedential, one-year joint licensing schemes including the JOL for the provision of Online and Mobile Music services to the public in the UK for private use.
- C The JOL has been referred to the Copyright Tribunal by the BPI and others, and Notices of Intervention in each of the References have been served by each of iTunes, the MNOs and the Academy.
- D MCPS, PRS, the Academy and the Settling Applicants have agreed to resolve their disputes in relation to the JOL and the References as set out below.

# It is agreed:

# 1 Definitions and Interpretation

1.1 Academy means the British Academy of Composers & Songwriters.

Applicant means any of the BPI, MusicNet (UK) Limited, Yahoo I (UK) Limited, AOL (UK) Limited, RealNetworks Limited, Napster LLC, Sony United Kingdom Limited, iTunes S.a.r.l and the MNOs.

BPI means the British Phonographic Industry Limited.

Copyright Tribunal means the statutory body established pursuant to Section 145 of the Copyright, Designs and Patents Act 1988 (as amended) and any statutory successor to that body.

IFPI means the International Federation of the Phonographic Industry.

iTunes means iTunes S.a.r.I.

iTunes' Disputed Contention has the meaning ascribed to it in clause 3.2 below.

Interveners means together iTunes and the MNOs.

**JOL** means the Licensing Scheme for the provision of online and mobile Music Services to the public for private use for the calendar year ending 31 December 2005.

Licensed Service has the meaning ascribed to it in the Scheme.

MCPS means Mechanical-Copyright Protection Society Limited.

MNOs means together 02 (UK) Limited, T-Mobile International (UK) Limited, Vodafone UK Content Services Limited and Orange Personal Communications Services Limited.

MNOs' Disputed Contention has the meaning ascribed to it in clause 3.1 below.

PRS means Performing Right Society Limited.

**References** means together the References commenced before the Copyright Tribunal and numbered 84/05, 85/05, 86/05, 87/05, 88/05, 89/05 and 90/05 of the JOL against MCPS and PRS.

**Scheme** means the joint MCPS and PRS scheme attached hereto and marked Schedule 1, save as amended by any decision of the Copyright Tribunal in the References in relation to the

iTunes' Disputed Contention, for iTunes, and in relation to the MNOs' Disputed Contention, for the MNOs.

Settlement Agreement means this Settlement Agreement including Schedules 1, 2 and 3.

Settling Applicants means together the BPI, iTunes and the MNOs.

1.2 Headings are for ease of reference only and shall not be taken into account in construing the Settlement Agreement.

#### 2 The Scheme

MCPS and PRS shall forthwith offer to iTunes (subject to the iTunes' Disputed Contention) and to the MNOs (subject to the MNOs' Disputed Contention), and shall offer to Full Members of the BPI from time to time during the term of the Scheme, to license each of them on the terms of the Scheme, and each of the Interveners shall agree to be bound by the terms of the Scheme, subject to the iTunes' Disputed Contention and the MNOs' Disputed Contention as the case may be.

# 3 Advertising Revenue

- 3.1 The Settlement Agreement shall be varied insofar as it relates to the MNOs as set out in this Clause 3.1. All terms of the Settlement Agreement shall apply to the MNOs as they apply to the BPI, save that the MNOs shall be entitled to continue as Interveners in the References solely for the purpose of contending before the Copyright Tribunal that revenue derived from advertising, sponsorship and commissions on or in connection with a Mobile Licensed Service should only be included within Gross Revenue as defined in the Scheme where that revenue is earned as a result of an advertisement, sponsorship or a click-through link located on a page of a Licensed Service from which the Mobile Licensed Service is actually offered (subject to apportionment where other services not requiring a licence are offered from the same page), and only where that Mobile Licensed Service is offered to the User for free ("the MNOs' Disputed Contention").
- 3.2 The Settlement Agreement shall be varied insofar as it relates to iTunes only as set out in this clause 3.2. All terms of the Settlement Agreement shall apply to iTunes as they apply to the BPI, save that iTunes shall be entitled to continue as an Intervener in the References solely for the purpose of contending before the Copyright Tribunal that revenue derived from advertising, sponsorship and commissions on or in connection with a Licensed Service should only be included within Gross Revenue as defined in the Scheme where that revenue is earned as a result of an advertisement, sponsorship or a click-through link located on a page of a Licensed Service from which the Licensed Service is actually offered (subject to apportionment where other services not requiring a licence are offered from the same page).

and only where the Licensed Service is offered to the User at a price which has been artificially depressed to reflect such revenue ("the iTunes' Disputed Contention").

# 4 Identity of the Licensee

The Licensee for the purposes of the Scheme shall be restricted to the Music Service Provider, as defined in the Scheme.

# 5 Term

The Scheme shall apply from 1 July 2006 to 30 June 2009. For the avoidance of doubt, the Scheme will terminate automatically on 30 June 2009.

#### 6 Podcasting

Podcasting Services shall be excluded from the Scheme.

#### 7 Other terms

- 7.1 Neither the Settlement Agreement nor the Scheme shall set any precedent for the licensing of musical works in the offline environment.
- 7.2 MCPS and PRS confirm their view that:
- 7.2.1 there is uncertainty as to how the UK online music industry may continue to develop in the future;
- 7.2.2 there are significant differences that exist between the offline and online exploitation of musical works (including both audio-only and audio-visual products); and
- 7.2.3 the Settlement Agreement and the Scheme should not, other than in relation to the References, set any precedent in relation to the licensing terms for the online or mobile provision of music.
- 7.3 The Settling Applicants do not agree with the view of MCPS and PRS referred to in clause 7.2 above.
- 7.4 The BPI shall forthwith notify IFPI and other national group members of IFPI of the contents of Clause 7.1 above.
- 7.5 MCPS and PRS shall forthwith notify BIEM and CISAC and those of their respective member societies involved in the administration of music rights of the contents of Clause 7.1 above.

- 7.6 MCPS and PRS confirm that Vodafone's Radio DJ service in its current form is a Webcasting Subscription Service for the purposes of the Scheme.
- 7.7 MCPS and PRS confirm that part A of the application form referred to in the definition of Agreement in the Scheme shall be drafted in good faith by MCPS and PRS by 4pm on 20 October 2006 and shall include only non-contentious, factual and current information about the Licensee and the Licensee's Music Service.
- MCPS and PRS on the one hand and the Settling Applicants on the other hand will negotiate in good faith what minima should apply to "long" (for example more than 10 minutes) and "short" (for example of less than one minute) tracks. Should these negotiations not be concluded by 10 November 2006, any of the MCPS, PRS or any of the Settling Applicants may refer the question as to what minima for "long" and "short" tracks (as referred to above) should apply for the period of the Scheme to an expert for determination. The identity of such expert shall either be agreed or, in the absence of such agreement within 7 days of the notice of referral, shall be determined by the Chairman for the time being of the Bar Council. MCPS, PRS and the Settling Applicants will seek to agree the procedure for the conduct of any expert determination. If such agreement is not reached within seven (7) days of the appointment of the expert, the expert shall have power to set the procedure for the expert determination.

# 8 Online Music Services prior to 1 July 2006

- 8.1 If and to the extent that any of the Interveners has in respect of its online and/or mobile use of any Musical Work prior to 1 July 2006, failed to make any payment and/or has failed to provide any reporting and accounting data due to MCPS and/or PRS pursuant to the terms of any licence granted to it by MCPS and/or PRS in relation to the provision of any of its Licensed Services, such Intervener shall:
- 8.1.1 within twenty-one (21) days of the Settlement Agreement provide MCPS and PRS with full and accurate reporting and accounting data for its online and/or mobile Music Services prior to 1 July 2006 in accordance with its existing licence;
- 8.1.2 pay any invoice issued by MCPS and/or PRS in relation to the provision of its Licensed Service prior to 1 July 2006 and calculated based on the data supplied in accordance with its existing licence within twenty-one (21) days of the date of such invoice; and
- 8.1.3 if any such invoice is not paid within twenty-one (21) days of the date it is issued, pay MCPS and/or PRS interest on the outstanding balance from time to time of such invoice calculated in accordance with the terms of its existing licence.

- 8.2 If and to the extent that any Intervener has not been licensed by MCPS and/or PRS in respect of its online and/or mobile use of any Musical Work for any period prior to 1 July 2006, such Intervener shall:
- 8.2.1 within twenty (20) days of the date this Settlement Agreement apply for a licence in accordance with the terms of the JOL in respect of such period prior to 1 July 2006 (and if it fails to do so, it shall be deemed to be subject to the terms of the JOL) and within forty-one (41) days of the date of this Settlement Agreement provide MCPS and PRS with full and accurate reporting and accounting data for its online and/or mobile music services prior to 1 July 2006 in accordance with the terms of the JOL;
- 8.2.2 pay any invoice issued by MCPS and/or PRS in relation to the provision of its online and/or mobile Music Services prior to 1 July 2006 and calculated based on the data supplied in accordance with the JOL within twenty-one (21) days of the date of such invoice; and
- 8.2.3 if any such invoice is not paid within twenty-one (21) days of the date of such invoice, pay MCPS and/or PRS interest on the outstanding balance from time to time of such invoice calculated in accordance with the terms of the JOL.
- 8.3 In calculating the amount to be invoiced pursuant to clauses 8.1.2 and 8.2.2 above, MCPS and/or PRS shall give credit for any payment received from the Intervener in relation to royalties due under its existing licence or the JOL as the case may be for any period prior to 1 July 2006.
- For the purposes of this clause 8, the expressions Music Services and Musical Work shall have the meanings respectively ascribed to them in the JOL.
- 8.5 Each of the Interveners agrees to report and account to MCPS and PRS in respect of its online and/or mobile use of any Musical Work from 1 July 2006 pursuant to and during the term of the Scheme.
- 8.6 Within twenty-one (21) days of the date of this Settlement Agreement, each of the Interveners shall provide MCPS and PRS with full and accurate reporting and accounting data for its online and/or mobile Music Services for each completed month since 1 July 2006 pursuant to the Scheme.
- 8.7 Each of the Interveners shall pay any invoice issued by MCPS and/or PRS in relation to the provision of its online and/or mobile Music Services during each completed month since 1 July 2006 within twenty-one (21) days of the date of such invoice.
- 8.8 If any such invoice, as referred to in Clause 8.7, is not paid within twenty-one (21) days of the date it is issued, the relevant Intervener shall pay MCPS and/or PRS interest on the

- outstanding balance from time to time of such invoice calculated in accordance with the terms of the Scheme.
- 8.9 The BPI shall use reasonable endeavours to encourage its members who are or have been Music Service Providers to comply with the obligations referred to in Clauses 8.1 to 8.8 above as if references to Interveners were to such members.
- 8.10 MCPS and PRS agree that they will consider in good faith any reasonable request made by an Intervener or a member of the BPI for a variation to the terms set out in this Clause 8, such variation to be set out in a confidential side letter between any such Intervener or member of the BPI and MCPS and PRS.

# 9 Further cooperation between the Parties

- 9.1 Given the fast-evolving nature of the UK online and mobile market, each of the Settling Applicants, MCPS and PRS agrees, in order to improve their understanding of the UK market, to facilitate future licensing arrangements, and to help to avoid a reference of the scheme(s) promulgated by MCPS and/or PRS in relation to the provision of online and mobile music in the UK after the expiry of the Scheme, the Settling Applicants, MCPS and PRS shall every quarter during the operation of the Scheme and on a strictly private and confidential basis:
- 9.1.1 exchange as between MCPS and PRS on the one hand and each of the Settling Applicants on the other hand information about relevant developments in the UK online and mobile music market, including as to licensed services, business models and pricing; and
- 9.1.2 unless otherwise agreed by MCPS and PRS on the one hand and each of the Settling Applicants on the other hand, meet on a quarterly basis to discuss the information exchanged pursuant to clause 9.1.1 above. For the avoidance of doubt, MCPS and PRS may be represented at such meetings by its management and/or by members of the Music Publishers' Association Board and BPI may be represented by its management and/or by members of the BPI Council or Rights Committee. The disclosure of confidential information by a Settling Applicant to any person other than the authorised management representatives of MCPS and PRS shall be subject to the consent of the disclosing Settling Applicant, such consent not to be unreasonably withheld, and the agreement of appropriate confidentiality arrangements, such agreement not to be unreasonably withheld by the disclosing Settling Applicant.
- 9.2 In the event that these meetings do not result in agreement between the Settling Applicants, MCPS and PRS as to the licence terms that should apply from 1 July 2009 in relation to the provision of online and mobile music in the UK, then MCPS and/or PRS shall produce and provide to the Settling Applicants on or before 1 May 2009 a proposed scheme or schemes which it and/or they propose should take effect from 1 July 2009. If MCPS and/or PRS fail to

provide such a scheme or schemes by that date, then and only then may the Settling Applicants or any of them refer the Scheme to the Copyright Tribunal.

#### 10 Universal Music Complaint

The BPI shall provide to MCPS and PRS on the date of the Settlement Agreement a letter signed by Universal Music Group withdrawing with immediate effect as against MCPS and PRS its complaint to the EU Commission (Case COMP C2/39.237 Universal Music v. collecting societies and publishers) in the form attached hereto and marked Schedule 2.

#### 11 Costs

- 11.1 Subject to Clause 11.2, MCPS, PRS, the Academy and the Settling Applicants agree that, as between themselves, each shall pay all of its own costs in relation to and/or incurred and/or for which it is already obliged to pay at the date of this Settlement Agreement as a result of the References. For the avoidance of doubt, to the extent that there are any existing Copyright Tribunal costs awards in the References as between MCPS, PRS, the Academy and the Settling Applicants, those awards shall not be enforced.
- 11.2 Clause 11.1 shall not apply to costs incurred by MCPS, PRS, the Academy, iTunes and the MNOs after the date of this Settlement Agreement in relation to the MNOs' Disputed Contention and/or iTunes' Disputed Contention and these costs shall be costs in the References.

# 12 Settlement Terms

- 12.1 The Settlement Agreement, as varied for iTunes pursuant to Clause 3.2 above and for the MNOs pursuant to Clause 3.1 above, is on the basis that:
  - (a) save for the purposes of the pursuit by the MNOs of the MNOs' Disputed Contention and the pursuit by iTunes of the iTunes' Disputed Contention, it is in full and final settlement of all claims and disputes between each of the Settling Applicants on the one hand and each of MCPS and PRS on the other hand howsoever arising directly or indirectly out of or in connection with the exploitation of Repertoire Works (as defined in the Scheme) prior to 1 July 2006 (where such exploitation would otherwise have fallen within the scope of the JOL) or the JOL and/or any of the References. For the avoidance of doubt, (i) nothing in this Settlement Agreement including this Clause 12.1(a) shall prevent any Settling Applicant from later referring to the Copyright Tribunal or otherwise challenging any podcasting scheme or licence promulgated or proposed (currently or in future) by MCPS and/or PRS; (ii) nothing in this clause 12.1(a) shall prevent the MCPS and PRS from enforcing their rights pursuant to this Settlement Agreement.

- (b) subject to Clause 9.2 above, no Settling Applicant shall refer, nor assist (save and to the extent required by any order of the Copyright Tribunal or a court of competent jurisdiction or by law so to do) a third party in referring, nor intervene in any reference made by a third party referring the Scheme;
- (c) save for the purposes of the pursuit by the MNOs of the MNOs' Disputed Contention and the pursuit by iTunes of the iTunes' Disputed Contention, and save and to the extent that it is required by any order of the Copyright Tribunal or a court of competent jurisdiction or by law so to do or under an existing contractual obligation to make any payment including to Richard Boulton, Derek Ridyard or their respective firms for work done in relation to the References after the date of the Settlement Agreement, no Settling Applicant shall in any way whether financially or otherwise assist any Applicant to pursue the References or any of them or any part of them against MCPS and PRS;
- (d) subject to the iTunes' Disputed Contention for iTunes and the MNOs' Disputed Contention for the MNOs:
  - the Settlement Agreement including the Scheme is a package deal, and taken
    in its entirety its terms are fair and reasonable; and
  - (ii) MCPS, PRS, the Academy, the BPI, the MNOs and iTunes shall advance and defend the position in the Copyright Tribunal (or any appeal) as described in Clause 12.1(d)(i), and shall contend that the Scheme represents the terms that should apply to all licensees for online and mobile music services within the scope of the Scheme for the purposes of the References;
- (e) Notwithstanding Clause 12.1(d) above, the execution of the Settlement Agreement shall not be treated as any acceptance by any of MCPS, PRS, the Academy or any of the Settling Applicants that any individual element of the Settlement Agreement including the Scheme is, viewed in isolation, fair and reasonable.
- (f) the BPI shall forthwith withdraw its Reference, its written submissions and its witness of fact evidence.
- (g) the MNOs and iTunes shall each forthwith amend their written opening submissions and witness of fact evidence, solely to the extent necessary to enable them to pursue the MNOs' Disputed Contention and iTunes' Disputed Contention as the case may be, and shall take no part in the References save and to the extent necessary to pursue the MNOs' Disputed Contention or the iTunes' Disputed Contention, as the case may be.
- (h) no order shall be sought from the Copyright Tribunal by the Settling Applicants, MCPS,
   PRS and the Academy in relation to the Settlement Agreement, and no order shall be

made as to costs as between MCPS, PRS, the Academy and the Settling Applicants save in relation to the costs referred to in Clause 11.2.

# 13 Jurisdiction

MCPS, PRS, the Academy and each of the Settling Applicants agree that any dispute arising out of or in connection with this Settlement Agreement shall be subject to the exclusive jurisdiction of the English Courts, save and to the extent that any such dispute is subject to the jurisdiction of the Copyright Tribunal.

#### 14 Public Statement

- 14.1 MCPS, PRS, the Academy, the BPI and/or the MNOs shall forthwith issue a press release in the form of the draft attached hereto and marked Schedule 3, and shall make no other public statement in relation to the Settlement Agreement including the Scheme unless it has the same or materially similar content, and iTunes shall make no public statement in relation to the Settlement Agreement including the Scheme.
- 14.2 MCPS and PRS shall provide a copy of the Settlement Agreement including the Scheme and any side letters to the Copyright Tribunal and to the Applicants which are not Settling Applicants, provided that such side letters shall be "class 2" Confidential Information as defined in the Mutual Non-Disclosure Agreement between the Applicants, MCPS, PRS and the Academy dated 28 April 2006 unless otherwise agreed by the parties to the relevant side letter.

# 15 Counterparts

This Settlement Agreement may be executed in any number of counterparts and by the parties to it on separate counterparts, each of which shall be an original, but all of which together shall constitute one and the same instrument. The Settlement Agreement is not effective until each party has executed at least one counterpart.

,	As duly authorised agent for and on behalf of
As duly authorised agent for and on behalf of	
By: [name & position]	By: [name & position]
As duly authorised agent for and on behalf of	As duly authorised agent for and on behalf of
British Academy of Composers & Songwriters	The British Phonographic Industry Limited
By: Man Line [name & position] (HAIR MAN	By:
As duly authorised agent for and on behalf of	As duly authorised agent for and on behalf of
iTunes S.a.r.l.	O2 (UK) Limited
By:	By:
As duly authorised agent for and on behalf of	As duly authorised agent for and on behalf of
T-Mobile International (UK) Limited	Vodafone UK Content Services Limited
Ву:	Ву:

[name & position]

[name & position]

As d	uly authorised agent for and on behalf of	As d	uly authorised agent for and on behalf o		
Mechanical-Copyright Protection Society Limited		Perf	Performing Right Society Limited		
By:		Ву:			
	[name & position]		[name & position]		
Aş dı	aly authorised agent for and on behalf of	As du	lly authorised agent for and on behalf of		
Britis	th Academy of Composers & Songwriters	The E	British Phonographic Industry Limited		
			90		
Ву:		Ву:	Ky m.		
	[name & position]		[name & position]		
	•		LOZ GADOME		
As du	ly authorised agent for and on behalf of	As du	ly authorised agent for and on behalf of		
iTune	s S.a.r.l.	O2 (U	K) Limited		
Ву:		Ву:			
	[name & position]		[name & position]		
As dul	y authorised agent for and on behalf of	As duly	y authorised agent for and on behalf of		
T-Mot	ile International (UK) Limited	Vodaf	one UK Content Services Limited		
Ву:		Ву:			
	[name & position]		[name & position]		

As duly authorised agent for and on behalf of  Mechanical-Copyright Protection Society Limited	As duly authorised agent for and on behalf of Performing Right Society Limited
By:	By:[name & position]
As duly authorised agent for and on behalf of British Academy of Composers & Songwriters	As duly authorised agent for and on behalf of  The British Phonographic Industry Limited
By:	By:
As duly authorised agent for and on behalf of iTunes S.a.r.l.	As duly authorised agent for and on behalf of O2 (UK) Limited
By: Dilly CUL EDDY CUE, DIRECTOR [name & position]	By:[name & position]
As duly authorised agent for and on behalf of  T-Mobile International (UK) Limited	As duly authorised agent for and on behalf of  Vodafone UK Content Services Limited  •
Ву:	By:

[name & position]

[name & position]

# 28/09 2006 THU 9:19 FAX 02087683150 Selsdon Park Hotel BC

As duly authorised agent for and on behalf of behalf of	As duly authorised agent for and on
Mechanical-Copyright Protoction Society Limited	Performing Right Society Limited
Ву:	Ву:
[name & position]	[name & position]
As duly authorised agent for and on behalf of behalf of	As duly authorised agent for and on
British Academy of Composers & Songwriters Industry Limited	The British Phonographic
ву:	Ву:
{rame & position}	[name & position]
As duly authorised agent for and on behalf of behalf of	As duly authorised agent for and on
Tunes S,a.r.l.	O2 (UK) Limited
зу:	By Jan
[name & position]	Russ Snaw Capability & Innovation Director
is duly authorised agent for and on behalf of ehalf of	As duly authorised agent for and on
-Mobile International (UK) Limited imited	Vodafone UK Content Services

As duly authorised agent for and on behalf of As duly authorised agent for and on behalf of Performing Right Society Limited Mechanical-Copyright Protection Society Limited Ву: ...... .,...... [name & position] [name & position] As duly authorised agent for and on behalf of As duly authorised agent for and on behalf of The British Phonographic Industry Limited **British Academy of Composers & Songwriters** By: By: [name & position] [name & position] As duly authorised agent for and on behalf of As duly authorised agent for and on behalf of O2 (UK) Limited iTunes S.a.r.l. By: [name & position] [name & position] As duty authorised agent for and on behalf of As duly authorised agent for and on behalf of Vodafone UK Content Services Limited T-Mobile International (UK) Limited By: By: [name & position]

11

ANDREW LEE Head of Legal

	•		
As dui	ly authorised agent for and on behalf of	As du	ly authorised agent for and on behalf o
Mech	anical-Copyright Protection Society Limited	Perfo	rming Right Society Limited
Ву:		Ву:	••••••
	[name & position]		[name & position]
As dul	y authorised agent for and on behalf of	As du	ly authorised agent for and on behalf of
Britisl	h Academy of Composers & Songwriters	The B	British Phonographic Industry Limited
Ву:		Ву:	
	[name & position]		[name & position]
As dul	y authorised agent for and on behalf of	As dul	y authorised agent for and on behalf of
Tune	s S.a.r.I.	O2 (U	K) Limited
Ву:	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Ву:	
	[name & position]		[name & position]
As dul	y authorised agent for and on behalf of	As dul	y authorised agent for and on behalf of
T-Mob	ile International (UK) Limited	Vodaf	one UK Content Services Limited
Ву:		Ву:	
	[name & position]		Nicholas Read, Director

08451215060 28. SEP. 2006 (THU) 10:21 ANANOVA TD As duly authorised agent for and on behalf of Ву:

# Schedule 1 - The Scheme

# LICENSING SCHEME FOR THE PROVISION OF ONLINE AND MOBILE MUSIC SERVICES TO THE PUBLIC FOR PRIVATE USE

# ONLINE AGREEMENT

- 1.1 The terms and conditions contained in the standard form Online Agreement ("the Online Agreement") which is set out in Appendix 1 will apply to an online and/or mobile Music Service where:-
  - (a) the service provider has applied to MCPS and PRS for that Agreement; and
  - (b) MCPS and PRS have granted that application in writing;
- 1.2 Application for an Online Agreement shall initially be in the form set out in Appendix 2, but MCPS and PRS shall be entitled to ask reasonable further questions to satisfy themselves of the eligibility of the applicant.

#### 2. AVAILABILITY OF ONLINE AGREEMENT

- 2.1 The Online Agreement is available to providers of online and mobile music services:
  - (a) who, in relation to such services, are the Music Service Provider (as that term is defined within the Online Agreement); and
  - (b) where the royalties payable (as defined in the Online Agreement) in relation to such music service(s) would be likely to be more than £200 per year.
- 2.2 Where a Quarterly Advance is payable under the Online Agreement, it shall be a minimum of £50.00 per Quarter.
- 2.3 Where 2.1(b) above does not apply, alternative licensing arrangements may be offered to the provider.

# TERRITORY

The territory covered by the Online Agreement is the United Kingdom of Great Britain and Northern Ireland, the Channel Islands and the Isle of Man. The position as to which additional countries (and in respect of which repertoire) can be included within the definition of 'Territory' may change throughout the Term. MCPS and PRS will inform the applicant for an Online Agreement as to which additional countries may be covered at the time of application.

# 4. LINKS TO UNLICENSED MUSIC

MCPS and PRS shall have the right to refuse to license any service which provides internet or mobile "links" to unlicensed music (whether in the form of recordings or notation, scores, lyrics, etc).

#### 5. DEPOSITS AND FINANCIAL GUARANTEES

MCPS and PRS may make the grant of an Online Agreement conditional upon the provision of such financial guarantees (by way of example only, deposits or advances) as are reasonably necessary to provide security against the risk that the members of MCPS and PRS may not receive such royalties as may be payable under the Online Agreement. Upon termination of an Online Agreement, MCPS and PRS shall release or repay any guarantee or similar security and shall repay any deposit or similar payment with accrued interest within 14 days of the

latest of (a) termination, or (b) the licensee having materially complied with all of its obligations under the Online Agreement or (c) resolution of any claim under any audit or other claim in relation to which notice was given to the Licensee prior to such termination.

For the avoidance of doubt, any guarantees under this provision are separate from the Quarterly Advance referred to in clause 6 of the Online Agreement and no interest shall be payable on any Quarterly Advance.

# 6. COMMENCEMENT AND TERM OF SCHEME

This scheme shall take effect on 1 July 2006 and shall continue until 30 June 2009.

# Appendix 1

#### THIS AGREEMENT IS MADE BETWEEN

MECHANICAL-COPYRIGHT PROTECTION SOCIETY LIMITED whose registered office is at 29-33 Berners Street London W1T 3AB ("MCPS") contracting for and on behalf of itself and for and on behalf of and as agent of its various Members and the Associated Societies (as defined below); and

PERFORMING RIGHT SOCIETY LIMITED whose registered office is at 29-33 Berners Street London W1T 3AB ("PRS") contracting for and on behalf of itself and for and on behalf of and as agent of the Associated Societies (as defined below); and

THE LICENSEE (as set out in schedule 6)

#### **WHEREAS**

- (A) The Licensee wishes to offer an online and/or mobile music service within the meaning set out in this Agreement;
- (B) PRS and MCPS have developed a joint licensing scheme for licensing musical works for use in such online and mobile music services;
- (C) PRS and MCPS have agreed to license musical works to the Licensee on the above basis and upon the terms and conditions contained in this Agreement.
- (D) Given the fast evolving nature of the UK online and mobile market, this Agreement shall operate only in relation to the period 1 July 2006 until 30 June 2009.

#### NOW IT IS HEREBY AGREED AS FOLLOWS

# 1. Definitions

"the Act"

means the Copyright, Designs and Patents Act 1988, as

amended from time to time.

"Agreement"

means these terms and conditions, the schedules to the terms and conditions and part A of the application form

completed by the Licensee.

"Alliance"

means MCPS-PRS Alliance Limited whose registered office is

at 29-33 Berners Street London W1T 3AB.

"Associated Society"

means each collecting society (or other body representing rights in Musical Works) with which MCPS and/or PRS has, from time to time, an agreement under which MCPS and/or PRS is authorised to grant licences in relation to the other society's (or body's) repertoire for the purpose of this Agreement PROVIDED THAT where such an agreement is only entered into after the commencement of the Term, a collecting society (or other body) shall only be regarded as an "Associated Society" for the purposes of this Agreement with effect from the date of signature of such agreement with

MCPS and/or PRS (as applicable).

# "Associated Society Member"

means any person, firm or company who or which has been notified, from time to time, as being a member of an Associated Society by the relevant Associated Society to MCPS and/or PRS.

#### "Audio-Visual Material"

means any specific presentation of Musical Works in conjunction with images, whether moving or still. For the avoidance of doubt (but without prejudice to the express restrictions contained in this Agreement), the following shall not be treated as Audio-Visual Material for the purpose of this Agreement:

- (a) the fact that ordinary web pages (or equivalent) are visible to the User while the User is listening to music; or
- (b) the fact that the media player used to play the music generates random visual images while the music is playing;

provided, in both cases, that the User would not reasonably be expected to associate the Repertoire Works being played with the images presented or think that there is any deliberate association by the Licensee of the Repertoire Works with such images.

#### "Commencement Date"

means the date specified in schedule 6.

#### "Commercial Work"

means any Repertoire Work other than:

- (a) one where the Member owning or controlling the copyright in such Repertoire Work has authorised MCPS to license it as so-called production or library music; or
- (b) a Commissioned Work, PROVIDED THAT for the purposes of clauses 3.2 and 4.5, a Commissioned Work shall not be excluded from such definition where the commissioning agreement does not authorise the Licensee to use the Commissioned Work in the context set out in those clauses.

#### "Commissioned Work"

means a Musical Work specially and expressly commissioned by the Licensee from composer/writer members of PRS and/or MCPS.

#### "Data Storage Device"

means any medium on which data can be stored (whether temporarily or permanently) whether existing now or invented in the future.

#### "Dramatico-Musical Work"

means any ballet, opera, operetta, musical, musical play or work of a similar nature.

# "Electronic Reporting Format"

means the format set out in schedule 4. If and when the DDEX format is agreed within the industry, the DDEX format shall replace the format set out in schedule 4 within 6 months of such agreement unless otherwise agreed between the parties, acting reasonably.

"Gross Revenue"

shall have the meaning set out in schedule 3.

"Licensed Services"

means the Music Services listed in schedule 6.

"Licensee"

means the party set out as such in schedule 6.

"Licensors"

means PRS and MCPS.

"Member"

#### means:

- (a) in the case of MCPS, each person, firm or company who or which, from time to time, has appointed MCPS as agent in relation to online exploitation either before or during the Term other than where such person, firm or company has opted not to participate in the licensing scheme pursuant to which this Agreement has been entered into, PROVIDED THAT a member who has so appointed MCPS after the commencement of the Term shall only be regarded as a member for the purposes of this Agreement with effect from the date on which the Member so appointed MCPS; and
- (b) in the case of PRS, any person, firm or company who or which, from time to time, pursuant to the Articles of Association of PRS has been admitted either before or during the Term as a member of PRS other than where such person, firm or company has reserved to himself the relevant rights pursuant to Article 7(cd) of the Articles of Association of PRS (or other equivalent article), PROVIDED THAT a member who has been so admitted after the commencement of the Term shall only be regarded as a member for the purposes of this Agreement with effect from the date of admission into PRS.

"Month"

means a calendar month.

"Musical Work"

means any musical work (as defined in the Act) and any lyrics or words written to be used with such musical work (if applicable). It includes any part of such a work.

"Music Service Provider"

means, the party which, in relation to a Music Service, most closely meets the following criteria:

- (a) contracts with the User in relation to the provision of the Licensed Service;
- (b) sets and controls the price the User pays;
- (c) can fully report on all elements of Gross Revenue (including relevant advertising revenues);
- (d) can fully report on all elements of music usage (or can procure such reporting);
- (e) controls how content is offered and bundled within the

#### Licensed Service;

 (f) carries out or authorises, on their instruction, the carrying out of the copyright restricted acts licensed under this Agreement.

# "Music Usage Information"

means the information referred to in the Electronic Reporting Format.

# "Music Videogram"

means any audio-visual production:

- (a) which has as the main feature of the soundtrack thereof a recording of a single Repertoire Work copies of which recording have been or are intended to be released as audio records for sale to the general public; and
- (b) the making of which was carried out by or on behalf of the record company releasing that recording or by or on behalf of the main artist(s) featured in that recording.

#### "Music Service"

means a music service falling within the definition set out in schedule 1.

#### "Permitted Excerpts"

refers only to Dramatico-Musical Works and means excerpts where the use of all such excerpts in any Audio-Visual Material complies with all the following limitations:

- (a) the total duration of the excerpts does not exceed 20 minutes;
- (b) the use is not a "potted version" of the Dramatico-Musical Work;
- (c) the use is not or does not cover a complete act of the Dramatico-Musical Work;
- (d) each excerpt is not presented in a "dramatic form" as defined below; and
- (e) as regards ballets specifically devised for television or excerpts from existing ballets, the total duration does not exceed five minutes.

A dramatic form shall be deemed to be created only by a performance in which there is a distinct plot depicted by actors and where the story of the Dramatico-Musical Work and/or its associated words is woven into and carries forward the plot and its accompanying action (a dramatic form shall not, for example, be deemed to be created by the use of costume, scenery, and/or any dance routine merely to provide an acceptable presentation of the work). For the purposes of this paragraph the word "actors" shall include actors, singers, mimics and/or puppets.

#### "Quarter"

means each of the periods from 1<sup>st</sup> January to 31<sup>st</sup> March, 1<sup>st</sup> April to 30<sup>th</sup> June, 1<sup>st</sup> July to 30<sup>th</sup> September, and 1<sup>st</sup> October to 31<sup>st</sup> December, throughout the Term.

"Quarterly Advance"

means the sum set out in schedule 6, excluding VAT (or other equivalent sales tax, as applicable).

"Relevant Party"

means a party which is involved in the provision of the Licensed Services, as set out in clause 15.2.

"Repertoire Work"

means each Musical Work the relevant copyright in which is owned or controlled, from time to time, in the UK by:

- (a) MCPS or a Member or an Associated Society or an Associated Society Member PROVIDED THAT (i) if one or more of those who own or control the copyright in a relevant Repertoire Work is not MCPS or a Member or an Associated Society or Associated Society Member, the expression "Repertoire Work" shall only apply to such interest in the Repertoire Work as is owned or controlled by MCPS or the Associated Society or the relevant Member or Associated Society Member, and (ii) it shall exclude any Musical Works that a Member of MCPS or an Associated Society has withdrawn or withheld from this Agreement; and
- (b) PRS or a Member or an Associated Society or an Associated Society Member PROVIDED THAT if one or more of those who own or control the copyright in a relevant Repertoire Work is not PRS or a Member or an Associated Society or Associated Society Member, the expression "Repertoire Work" shall only apply to such interest in the Repertoire Work as is owned or controlled by PRS or the Associated Society or the relevant member or Associated Society Member.

For the avoidance of doubt, if a Musical Work is a Repertoire Work in relation to one Licensor and not the other then it remains a Repertoire Work under this Agreement in relation only to the licence granted by that Licensor.

#### "Reporting Date"

#### means:

- (a) where the Licensee is undertaking monthly accounting in accordance with clause 6, 5.00pm on the Friday following the Week to which the reporting relates; or
- (b) where the Licensee is undertaking quarterly accounting in accordance with clause 7, one month following the Quarter to which the Music Usage Information relates.

"Royalty Fee"

means the royalties payable as set out in schedule 2.

"Server Territory"

means the European Economic Area, Switzerland, the United States of America, Canada and such other territories as may be agreed in writing by the parties.

"Term"

means the period starting on the Commencement Date and ending upon the date set out in clause 13.1 (unless terminated earlier under clauses 4.6, 13.2, 13.3, 13.4 or 13.5).

"Territory"

means the UK, and those territories (if any) listed in schedule 6 (subject always to clause 4.6 and schedule 2, paragraph 5) and such other countries as the Licensors and the Licensee

may agree in writing.

"UK" means the United Kingdom of Great Britain and Northern

Ireland and the Channel Islands and the Isle of Man.

"User" means a natural person in the Territory who receives the

Licensed Services for their own private and non-commercial

use.

"VAT" means value added tax pursuant to the Value Added Tax Act

1994 and each like tax imposed in addition to or in

substitution therefor.

"Week" means a week starting on Monday and ending on Sunday.

#### 2. Grant of Licence

2.1 Subject to and conditional upon compliance with the terms and conditions contained herein and in particular subject to the exclusions and restrictions set out in clauses 3 and 4, MCPS grants the Licensee a non-exclusive licence to do the following during the Term:

- (a) to reproduce Repertoire Works on servers within the Server Territory for the purpose
  of transmitting the same to Users (for the User's own private and non-commercial use)
  by means of the Licensed Services; and
- (b) where the Licensed Services expressly authorise the temporary or permanent reproduction of Repertoire Works on Users' Data Storage Devices, to cause such copies to be made in the Territory for the User's own private and non-commercial use; and
- (c) to authorise the reproduction of "pre-loaded" copies of Musical Works on Data Storage Devices and distribute such Data Storage Devices to the public within the Territory in an encrypted or other protected form within consumer electronic equipment where the Repertoire Works are solely intended to be accessible at a later date only to Users (for their own private and non-commercial use) through the Licensed Service(s). For the avoidance of doubt, the royalties set out in schedule 2 (including the Minimum Royalties) shall be payable in respect of Musical Works "pre-loaded" in accordance with this sub-clause 2.1(c). However, the royalties in respect of such "pre-loaded" Musical Works shall not accrue until the User has become able to access and play such Musical Works. This is without prejudice to the obligation of the Licensee to pay the Quarterly Advance.

For the avoidance of doubt (but subject to all terms of this Agreement, in particular clause 3.3), it is intended that this Agreement licenses all reproductions of Repertoire Works necessary in the ordinary operation of the Licensed Services.

- 2.2 Subject to and conditional upon compliance with the terms and conditions contained herein and in particular subject to the exclusions and restrictions set out in clauses 3 and 4, PRS grants the Licensee a non-exclusive licence, during the Term, to communicate to the public (as that term is defined in the Act) and to authorise the communication to the public of Repertoire Works within the Territory solely as part of and for the purposes of the provision of the Licensed Services.
- 2.3 The above licences shall apply where Repertoire Works are used in audio-only material and, subject to clause 3.2, Audio Visual Material.
- 2.4 The Licensors will not unreasonably refuse or delay any request by the Licensee to include further services of the Licensee within the scope of this Agreement provided that:
  - (a) they are a Music Service; and

- (b) the Licensee is, in relation to that service, the Music Service Provider; and
- (c) they are otherwise within the scope and limitations set out in this Agreement.
- 2.5 The provisions of clauses 3.2, 3.6, 4.1, 4.2, 4.4 and 4.5 shall not apply to a particular Repertoire Work where the owner of the relevant rights in such Repertoire Work has granted permission to the Licensee for the use of that Repertoire Work on the Licensed Services in the manner described under the relevant clause on such terms and conditions (including, if required, the payment of royalties or fees in addition to those specified under this Agreement) as the owner thinks fit.
- 2.6 The licences granted in clauses 2.1 and 2.2 above shall not apply to any Licensed Service which knowingly or recklessly provides internet or mobile "links" to music which requires a licence, but is unlicensed (whether in the form of recordings or notation, scores, lyrics, etc). The inclusion of such links on a Licensed Service shall constitute a material breach of this Agreement (which is capable of remedy).
- 2.7 Where the Licensee wishes to launch a service where music is the primary theme of such service, but such service does not fall within the definition of Music Service as set out schedule 1 to this Agreement, then the Licensors agree to enter into good faith discussions with the Licensee concerning the inclusion of such new service within the scope of this Agreement and the appropriate royalty rate and minima which shall be applicable.

#### 3. Exceptions and Limitations

- 3.1 The licences granted under clause 2 of this Agreement are valid only insofar as:
  - (a) the Licensed Services are Music Services; and
  - (b) the Licensee is the Music Service Provider in relation to the Licensed Services.
- 3.2 Subject to clause 2.5, the incorporation of Commercial Works into Audio-Visual Material is only licensed under this Agreement where the Audio-Visual Material consists of:
  - (a) a Music Videogram; or
  - (b) a live concert performance or a film of a live concert performance by the artist performing that particular Commercial Work; or
  - (c) subject to clause 4.5, such Commercial Works being combined with photographs or other images relating to the artist performing the Commercial Work or the composer of the Commercial Work; or
  - (d) subject to clause 4.5, an interview with an artist, composer, producer or other person involved in the creation, performance or production of music where the Commercial Work(s) used are associated with the interviewee(s); or
  - (e) where permitted under clause 4.1, a performance of Permitted Excerpts of the Dramatico-Musical Work of which the Commercial Work forms part.

For the avoidance of doubt, such Audio-Visual Material as is licensed under 3.2(a) to (e) above is only licensed insofar as it is reproduced or communicated to the public via the Licensed Services.

3.3 For the avoidance of doubt, this Agreement does not grant any "synchronisation licence" (to the extent that such a licence may be required by the Licensee) covering the initial fixation of Repertoire Works in combination with visual images to create and produce Audio-Visual Material.

- 3.4 For the avoidance of doubt and except as specifically provided for in clause 2.1(c), the licences granted under clause 2 of this Agreement shall not authorise the manufacture or distribution of physical products containing Repertoire Works, such as (without limitation) the ordering of compact discs (or any other type of physical media) via a Music Service, but which are distributed by mail.
- 3.5 For the avoidance of doubt, the licences granted under clause 2 of this Agreement shall not extend to the public performance (as that term is used in the Act) of Repertoire Works, whether as part of the Licensed Services or otherwise.
- 3.6 Subject to clause 2.5 and paragraph 5 of schedule 3, the licence granted under clause 2.1 of this Agreement shall not permit the use of Repertoire Work(s) with any advertising or sponsorship of whatsoever nature where:
  - (a) such Repertoire Work(s) are incorporated into such advertising or sponsorship; or
  - (b) such Repertoire Work(s) are otherwise presented in such a way that a reasonable person might associate the Repertoire Work(s) with the advertising or sponsorship.
- 3.7 Subject to paragraph 5 of schedule 3, for the avoidance of doubt (but without prejudice to the generality of clause 3.6), the licences granted under this Agreement shall not apply to any Repertoire Work(s) made available for the purpose of (whether in whole or in part):
  - (a) directly or indirectly encouraging the User to purchase or obtain goods or services of whatsoever nature (other than music via the Licensed Service); or
  - (b) promoting the branding of the Licensee, any affiliate of the Licensee or any third party; in such a manner that:
  - (i) one or more particular Repertoire Works, composers or writers are associated with such promotion; or
  - (ii) a reasonable person might assume that there was an association between particular Repertoire Works, composers or writers and such promotion.
- 3.8 The licence granted under clause 2.1 shall not apply to graphic copies (meaning, without limitation, copies of lyrics, notation or scores) of Repertoire Works. For the avoidance of doubt, the licences granted under this Agreement shall not apply to any "karaoke" service within a Music Service.
- 3.9 The licence granted under clause 2.2 shall only apply to a Repertoire Work communicated to the public as part of a Licensed Service where the Licensee has the benefit of a valid licence for or a right to make a reproduction of that particular Repertoire Work and for that particular form of exploitation via the Licensed Service either pursuant to this Agreement or otherwise. By way of example only, the licence granted by MCPS under clause 2.1 does not cover the copying of Repertoire Works in an advertisement (see clause 3.6 above). However, clause 3.6 does not apply to the licence granted by PRS under clause 2.2. Therefore, pursuant to this clause 3.9, the licence granted by PRS under clause 2.2 would not apply (in the context of this example) unless the Licensee has the benefit of a right to copy the Repertoire Work for use in that advertisement (and on the Licensed Service(s)).
- 3.10 For the avoidance of doubt, this Agreement grants no licence whatsoever in relation to Repertoire Works which are made available by the Licensee outside of the Licensed Service.

However, such other exploitation of Repertoire Works may fall within the scope of other licensing schemes operated by the Licensors, details of which shall be made available to the Licensee on request.

#### 4. Further Restrictions

- 4.1 Subject to clause 2.5, where any Repertoire Work forms part of any Dramatico-Musical Work, the licence granted under clause 2.1 shall not apply, in relation to Audio-Visual Material, to the reproduction of:
  - (a) the whole Dramatico-Musical Work; or
  - (b) any excerpt(s) from such Dramatico-Musical Work unless all of the following circumstances apply:
    - that which is copied or communicated to the public via the Licensed Services under this Agreement contains only excerpt(s) within the definition of Permitted Excerpts; and
    - (ii) neither of the Licensors has notified the Licensee in writing that their Member or the Associated Society Member objects to the reproduction of any such Repertoire Work.
- 4.2 Subject to clause 2.5, where any Repertoire Work forms part of any Dramatico-Musical Work, the licence granted under clause 2.1 shall not apply, in relation to material other than Audio-Visual Material, to the reproduction of the whole or substantially the whole Dramatico-Musical Work unless:
  - (a) the Licensee has specifically notified the Licensors that it wishes to reproduce the whole or substantially the whole work; and
  - (b) the Licensors have notified the Licensee that all relevant Members consent to such reproduction.

For the purposes of this clause 4.2, the expression "Dramatico-Musical Work" shall include any version of such work (with or without cuts, additions, interpolations or the like) which has —been publicly performed. Furthermore, for the avoidance of doubt, substantially the whole work shall be deemed to be reproduced where all or nearly all the individual songs or other music included in the work are reproduced.

- 4.3 In any event, any licence hereunder only applies to the relevant Repertoire Works and not (by way of example only) to any underlying dramatic or literary work which forms part of the Dramatico-Musical Work or which such Dramatico-Musical Work is based on or uses.
- 4.4 Subject to clause 2.5, the licences granted under this Agreement shall not extend to or permit any adaptation of any Repertoire Work to be copied or communicated to the public as part of a Licensed Service unless the relevant Member has consented to such adaptation. By way of example only, this applies to:
  - (a) any sampling (meaning the taking of part of the music and/or lyrics of a Repertoire Work and incorporating such part into another Musical Work) or the communication to the public or reproduction in the form of a sample of such part of a Repertoire Work; or
  - using with music lyrics other than those written to be used with the music or authorised for use with the music; or
  - (c) using with lyrics music other than that written to be used with the lyrics or authorised for use with the lyrics.

However, subject always to clause 4.10 and provided that such alterations do not amount to an adaptation of a Repertoire Work and do not contravene clause 4.5, then this Agreement shall apply in relation to Repertoire Works that have been modified (including music and/or lyrics) for the purpose of satisfying the requirements of the relevant recording.

For the avoidance of doubt, the Licensors acknowledge, for the purposes of the restriction set out in this clause 4.4, that the production and inclusion as part of the Licensed Services of audio clips of Repertoire Works of up to 30 seconds (or, for "classical" genre Repertoire Works, up to 1 minute) duration to promote the supply of music via the Licensed Services does not of itself constitute an adaptation or sample.

- 4.5 Subject to clause 2.5, the licences granted under this Agreement shall not extend to:
  - (a) the reproduction or communication to the public of any Commercial Work or part thereof in the form of a parody or burlesque of any Commercial Work or of any composer or writer of any Commercial Work or any band or other group of artists which includes any composer or writer of any Commercial Work; or
  - (b) the use of any Commercial Work in any context which the Licensee ought reasonably to consider as being likely to be insulting or detrimental to the composer featured on the commercially released sound recording of the music or the relevant Member or Associated Society Member.
- 4.6 Any additional limitations in relation to the Associated Societies' rights to grant the licences set out in clause 2 of this Agreement which have been notified to the Licensors shall be notified to the Licensee in writing (which may include by email) and shall be binding no less than 10 days following such notice. Where any restriction of a material nature is added, the Licensee shall have the right to terminate this Agreement by giving written notice to the Licensors.
- 4.7 All rights not specifically granted under this Agreement are hereby reserved.
- 4.8 This Agreement only covers Repertoire Works. It does not extend to other rights or interests, including (by way of example only), rights in sound recordings, films, dramatic works, performers' rights or rights in performances. The Licensee is required to obtain the appropriate waivers, consents and/or licences from the person(s) owning or controlling rights in relation to sound recordings containing Repertoire Works or performers of that Repertoire Work.
- 4.9 It is the responsibility of the Licensee to obtain all necessary licences in relation to any Musical Work which is not, or to the extent that it is not, a Repertoire Work, and no licence is granted under this Agreement in relation thereto.
- 4.10 Nothing in this Agreement affects the moral rights of authors of Repertoire Works whether subsisting in the UK or any other territory.

# 5. Payment and Accounting

- 5.1 Where the Licensors are satisfied that:
  - (a) the projected Gross Revenue of the Licensee during the first year of the Agreement is over £500,000 (excluding VAT or other equivalent sales tax), such figure to be increased each calendar year in accordance with the Retail Price Index change over the previous calendar year; and
  - (b) there is no adverse credit risk causing the Licensors to have reasonable concerns over the Licensee's ability to pay royalties due under this Agreement; and
  - (c) the Licensee is capable of accurately reporting and paying royalties on a monthly basis in accordance with the provisions of this Agreement;

then the Licensee shall pay royalties in accordance with clause 6 below.

5.2 Otherwise, the Licensee shall pay royalties in accordance with clause 7 below.

- 5.3 All licence fees and payments referred to in this Agreement are subject to VAT or other equivalent sales tax. The Licensee shall pay to the Licensors VAT or other equivalent sales tax (if applicable) at the rate or rates from time to time in force on any sums payable under this Agreement.
- 5.4 Except as expressly set out in this Agreement, no deduction in respect of any tax, or any other deduction or set-off of whatsoever nature, shall be made in calculating or paying any sum due under this Agreement.
- 5.5 Notwithstanding the provisions of clauses 6 and 7 below, the Licensors confirm and warrant that the Alliance is authorised to receive all payments under this Agreement as agent on behalf of the Licensors, each of the Members and, subject to clause 4.6, the Associated Societies.
- 5.6 Without prejudice to any other right or remedy of the Licensors, and without imposing an obligation to accept late payment, where any fees payable under this Agreement are not paid by the due date (or the date on which such fees should ordinarily have been paid in circumstances where the Licensors have been unable to submit an invoice) due to default of the Licensee, the Licensee shall (if required by the Licensors) pay interest on such late payment calculated on a daily basis at an annual rate of 3% over the base rate, current from time to time, of National Westminster Bank Plc payable from the date on which the payment should have been made to the date on which the payment was made.
- 5.7 All payments made under this Agreement shall be in Sterling unless otherwise agreed by the parties in writing. Where it is necessary to convert an amount payable to Sterling from another currency, the exchange rate used shall be the Financial Times closing mid market rate on the last working day of the Quarter to which the payment relates. The Licensee shall pay all bank charges on transfers of sums payable by the Licensee to the Licensors (or the Alliance pursuant to clause 5.5).
- 5.8 The address for delivery of the royalty statement referred to in clauses 6.2 and 7.5 is Online Licensing Team, MCPS-PRS Alliance-Limited, Copyright House, 29-33 Berners Street, London, W1T 3AB.
- 5.9 In relation to Permanent Download Services and LD/ODS Services only, if and insofar as, (i) a Musical Work is not in copyright in the relevant part of the Territory, or (ii) it is not a Repertoire Work, credit shall be given by either or both of the Licensors (as the case may be) for any overpayment of royalties. The parties shall discuss in good faith the mechanism for calculating and granting any rebate for Musical Works which are not Repertoire Works. Unless and until agreed otherwise, the terms set out in this Agreement shall apply, it being acknowledged by the Licensors that it may then be necessary to make adjustments to payments that have already been made by the Licensee to the Licensors as from the Commencement Date so as to allow for royalties paid on Musical Works that are not Repertoire Works.
- 5.10 Notwithstanding the above, for Music Services where a significant proportion of Musical Works accessed by Users are either no longer in copyright in the relevant part of the Territory or are otherwise not Repertoire Works (by way of example, services specialising in classical music), the Licensors shall agree an appropriate deduction to the Gross Revenue and Minimum Royalties to apply as from the Commencement Date. This deduction shall be reviewed every six months.

# 6. Fees and Payment - Monthly Accounting

- 6.1 In consideration of the licences and authorisations granted under this Agreement, the Licensee shall pay to the Licensors the Royalty Fee.
- 6.2 Within 8 working days of the end of each Month, the Licensee shall send to the Licensors a fully and accurately completed royalty statement (in the form attached in schedule 5).

- 6.3 The Alliance shall, on behalf of the Licensors, each of the Members and, subject to clause 4.6, the Associated Societies, raise an invoice for the due amount, and the Licensee shall pay such invoice by electronic bank transfer (or, if agreed by the Licensee, direct debit) within 10 days of the invoice date.
- Where, in relation to any particular Month, the Licensee fails to provide by the required date the information required under the Agreement to calculate the Royalty Fee, then the Licensors shall be entitled to fix the Royalty Fee based on (a) the Royalty Fees payable in previous Months and (b) any other relevant factors which could reasonably lead the Licensors to believe that the Royalty Fees payable would be materially different from those paid or payable in previous Months.

# 7. Fees and Payment – Quarterly Accounting

- 7.1 In consideration of the licences and authorisations granted under this Agreement, the Licensee shall pay to the Licensors:
  - (a) the Quarterly Advance; and
  - (b) subject to clause 7.2, the Royalty Fee.
- 7.2 The Quarterly Advance is recoupable against the Royalty Fee. If any part of the Quarterly Advance remains unrecouped at the end of a Quarter, such unrecouped amount may be carried over to subsequent quarters, but for the avoidance of doubt, shall not (except as set out in clause 7.3 below) have the effect of reducing any subsequent Quarterly Advance. If any part remains unrecouped upon termination or expiry of the Agreement, the Licensors shall return such part to the Licensee.
- 7.3 The Quarterly Advance shall be reviewed at the end of each calendar year. It may also be reviewed during a calendar year if the royalties calculated for any particular Quarter under schedule 2 differ from the Quarterly Advance by 20% or more.
- 7.4 Where, in relation to any particular Quarter, the Licensee fails to provide by the required date the information required under the Agreement to allow the calculation referred to in clause 7.1(b), then the Licensors shall be entitled to fix the Royalty Fee based on (a) the Royalty Fees payable in previous Quarters and (b) any other relevant factors which could reasonably lead the Licensors to believe that the Royalty Fees payable would be materially different to those paid or payable in previous Quarters.
- 7.5 The Quarterly Advance shall be payable by direct debit on the first day of each Quarter. Where the Agreement begins (and/or a Licensed Service comes into operation) within a Quarter, the first Quarterly Advance payment shall be pro-rated (according to the Commencement Date) and shall be payable upon the later of signature of this Agreement and the date on which the first Licensed Service is made available to Users for the first time.
- 7.6 Within 15 working days of the end of each Quarter the Licensee shall provide to the Licensors a fully and accurately completed royalty statement (in the form attached in schedule 5).
- 7.7 The Alliance shall, on behalf of the Licensors, each of the Members and, subject to clause 4.6, the Associated Societies, raise an invoice for the due amount, and the Licensee shall pay such invoice by electronic bank transfer (or, if agreed by the Licensee, direct debit) within 21 days of the invoice date.

# 8. Supply of Information

8.1 In relation to any and all Repertoire Works reproduced and communicated to the public (or otherwise accessed through a Licensed Service if from "pre-loaded" copies pursuant to clause 2.1(c)) under this Agreement via all Licensed Services, the Licensee will deliver the Music Usage Information to the Licensors or to the Licensors' duly authorised agent (details of which will be provided to the Licensee) in the Electronic Reporting Format weekly (where the

Licensee is undertaking monthly accounting) or quarterly (where the Licensee is undertaking quarterly accounting) by the Reporting Date.

- 8.2 The Licensee must also supply the Licensors with any further information or documentation in its possession, power, custody or control (and use its reasonable endeavours to supply the Licensors with any further information or documentation not in its possession, power, custody or control) reasonably requested by the Licensors at any time, in order to enable the Licensors to verify the Repertoire Work(s) which have been reproduced or distributed via all Licensed Services.
- 8.3 Where any or all of the Licensed Services are accessible by Users only on payment of subscription or other similar payment (or access is otherwise limited or controlled in some way), the Licensee shall, upon request of the Licensors, use reasonable endeavours to ensure that such Licensed Services are at all times accessible by the Licensors (and the Alliance) free of charge for the purposes of the Licensors verifying that the Licensee is acting in accordance with this Agreement. For the avoidance of doubt, the Licensee shall not be required to provide free access to downloads under a Permanent Download Service or to provide free access to a mobile phone network or internet access.
- 8.4 The Licensee acknowledges that the Licensors have a responsibility to maximise the efficiency of their reporting to their Members and the Associated Societies. Therefore, if the Licensors wish to make any reasonable upgrade or alteration of whatsoever nature to the Music Usage Information or data specification referred to in schedule 4 during the Term, the Licensee agrees to use its reasonable endeavours to implement the changes required as soon as is reasonably practicable, PROVIDED THAT the Licensors shall not request that the Licensee implements the change in less than six months from the making of the request, and in each case of a change the following procedures shall apply:
  - (a) the Licensors shall give full details thereof in writing to the Licensee;
  - (b) the Licensee will respond in writing within 4 weeks of receipt of the request, stating the date by which it commits to comply with the change and the Licensors shall provide full assistance to the Licensee in order to assist the Licensee in complying with the change; and

the parties will then finalise the details and undertake tests to ensure that the change operates satisfactorily within the terms of this Agreement, satisfactory operation of which will be deemed acceptance of the change and schedule 4 and/or the definition of Music Usage Information will be amended or replaced accordingly. For the purpose of this clause, in determining what change may be reasonable, regard shall be had to the DDEX project.

- 8.5 Without prejudice to any right in law that the Licensors may have to obtain such information, the Licensee shall not be obliged to provide to the Licensors any information which identifies Users or which otherwise constitutes "personal data" as defined in the Data Protection Act 1998. For the avoidance of doubt, the Licensee must still provide all required Music Usage Information (or other information to be provided under this Agreement), but is entitled to remove any element of it which reveals the identity of Users or otherwise causes it to include or constitute "personal data".
- 8.6 The Licensee shall notify the Licensors in writing in advance of all arrangements which it enters into for "pre-loading" content on Data Storage Devices pursuant to clause 2.1(c), such notification to include details of the relevant Data Storage Devices.

#### 9. Late Reporting

- 9.1 The following provision applies where the Licensee has:
  - failed to deliver prior to the required date Music Usage Information for the Licensed Service; or

- (b) delivered such Music Usage Information prior to the required date, but it contains any omission or error of whatsoever nature (by way of example only, Repertoire Works having been omitted therefrom or incorrectly or misleadingly named, or the duration of any Repertoire Works having been under-reported) which is material and the Licensee has failed to give notice in writing to the Licensors correcting the omission or error by the date upon which such reporting should have been provided.
- 9.2 In such circumstances (and without prejudice to any other rights which either or both Licensors may have against the Licensee) if one or both of the Licensors has distributed some (or all) royalties received (excluding any commission) to Members and/or Associated Societies based on the usage data supplied, but, as a result of clause 9.1(a) or (b) applying, the royalties distributed do not accurately or fairly represent that which each of the Members and/or Associated Societies should have received (when considering the true usage of Repertoire Works), then the Licensee shall pay the fees set out in clauses 9.3 and 9.4 (in addition to those set out in clauses 6 and 7).
- 9.3 Where clause 9.2 applies, separately in relation to each relevant Repertoire Work either omitted from the relevant Music Usage Information or as regards which there was a material omission or error of whatsoever nature or in relation to which the relevant Music Usage Information was not delivered, the additional fees shall be calculated at the rate equivalent to that which has been or will be paid by MCPS and/or PRS (as applicable) to their members (or associated societies) in relation to the Music Usage Information which was submitted prior to the required date.
- 9.4 Where clause 9.2 applies, the Licensee will also pay interest on the additional fees as referred to in clause 9.3 above computed in accordance with clause 5.6 and calculated from the date on which MCPS and/or PRS (as applicable) first made a distribution to their members (or associated societies) in relation to the relevant period to the date on which the Music Usage Information was received by the Licensors or the date on which the Licensors received written notice of the relevant error, as the case may be.
- 9.5 The provisions of this clause apply notwithstanding any other provision of this Agreement, but are without prejudice to any other right which MCPS and PRS have in relation to any failure to submit Music Usage Information fully or accurately completed within the time stipulated in clause 8.1.

#### 10. Credits and Notices

The Licensee shall include on each of the Licensed Services:

- (a) the logos of PRS and MCPS; and
- (b) details of the following website and, where practical, hypertext links to it: http://www.mcps-prs-alliance.co.uk; and
- (c) where reasonably practicable, the name of the composer and publisher of the Repertoire Works provided via the Licensed Services; and
- (d) a notice explaining that use of the musical works is subject to restrictions and that a summary of these restrictions may be obtained by accessing the Licensors' website through the link referred to above.

Approval of the Licensors shall be deemed to be given to the positioning of the above credits and notices within the terms and conditions of the Licensed Service(s) where it is not reasonably practical to position such information elsewhere.

#### 11. Auditing

11.1 The Licensee shall keep and make available for inspection upon reasonable notice (and shall procure that each Relevant Party keeps and makes available for inspection upon reasonable

notice), both during and for twelve months after termination of this Agreement, proper, detailed books and records relating to (a) the use of all Musical Works and (b) any income or other consideration received by or on behalf of the Licensee in relation to the Licensed Services, together with any supporting documentation relating thereto covering the period up to six years prior to the date of notification of audit.

- 11.2 For the purposes of this clause 11, the Licensee shall allow upon reasonable notice (and shall procure that each Relevant Party shall allow) access to its premises to inspect relevant accounting records, but not more than once per annum. The duly authorised representatives (who shall be external qualified accountants or auditors unless otherwise agreed between the parties) of the Licensors shall have such access to the Licensee's premises and shall be entitled to inspect, make extracts and take copies of any of the information and/or documentation available and to carry out such work as is, in their reasonable opinion, considered necessary to verify compliance with the provisions of this Agreement.
- 11.3 If tests under any audit and verification process indicate under-payment of the correct Royalty Fee during the period under audit, then, without prejudice to the Licensors' other rights under this Agreement, the Licensee shall pay the amount of the underpayment plus interest based on the period from which the correct fee should have been paid to the Licensors to the date when it was actually paid (at the rate set out in clause 5.6).
- 11.4 If any audit and verification process discloses (a) under-payment of more than 7.5% of the correct Royalty Fee during the period under audit and/or (b) failures to report correctly (so as to affect a distribution by the Licensors to their Members) amounting to at least 7.5% of the music usage during the period under audit, then, without prejudice to the Licensors' other rights under this Agreement, the Licensee shall pay, in addition to the payment referred to in clause 11.3, the Licensors' reasonable costs of such audit and verification within 28 days of receipt of the Licensors' VAT invoice therefor.
- 11.5 If tests under any audit and verification process indicate over-payment of the correct Royalty Fee during the period under audit, then the Licensors shall, as soon as is reasonably practical, pay the amount of the overpayment back to the Licensee (but, for the avoidance of doubt, no interest shall be payable unless the overpayment is a result of an act or omission of the Licensors (in which case interest shall be payable at the rate set out in clause 5.6)). However, where the overpayment does not result from an act or omission of the Licensors and the Licensors have already distributed such overpayment to their Members and/or Associated Societies, the Licensors shall be entitled to deduct its reasonable internal and/or external costs in administering the payment back of the overpayment.
- 11.6 The Licensors shall not (and shall procure that their representatives shall not), without the Licensee's written consent, disclose to any third party any confidential information of the Licensee (so long as it remains confidential) received in the course of an audit carried out under this clause 11, save that such confidential information may be disclosed to the Licensors' directors, board sub-committee members, officers, employees and professional advisors (solely where such persons are under a duty of confidentiality in relation to information so received and the Licensors shall be liable to the Licensee in respect of any breach of such confidentiality obligation) solely for purposes connected with this Agreement.
- 11.7 For the avoidance of doubt, books, records and accounting records as referred to in clauses 11.1 and 11.2 above shall include data, information and records held on computers.
- 11.8 The Licensee shall, if requested by the Licensors, provide a statement from its auditors (but not more than once per year) confirming that the financial information submitted by the Licensee for the relevant requested period is in accordance with the actual Gross Revenue (as defined under this agreement) for that period.

#### 12. Security and Encryption

12.1 Unless agreed otherwise, the Licensee will utilise or require the utilisation of an industry security standard which is developed and is available for use in the protection of Repertoire Works. Until such time, the Licensee must use its reasonable endeavours to prevent unauthorised copying and/or the unauthorised issuing of copies of Repertoire Works by whatever technical means are practicable. Upon request, the Licensee will inform the Licensers concerning its activities in relation to fulfilling this obligation.

- 12.2 Save as may be permitted by law, the Licensee agrees it shall not (and shall procure that any Relevant Party shall not) attempt to:
  - remove or alter any rights management or identifier information that may be associated with any Repertoire Works; or
  - (b) circumvent any technical measures associated with any Repertoire Work which are designed to prevent or restrict the unauthorised use of any Repertoire Work.

# 13. Termination and Expiry

- 13.1 This Agreement shall expire on 30 June 2009 unless terminated earlier by written agreement or in accordance with the terms of this clause 13.
- 13.2 This Agreement may be terminated by the Licensee, by giving not less than three months written notice to the Licensors or upon written notice in circumstances where the Licensee is ceasing to engage in activities covered by this Agreement.
- 13.3 Each party shall have the right to terminate this Agreement by notice forthwith where the other party:
  - (a) commits a material breach of this Agreement which is capable of remedy and fails to remedy such breach within 14 clear days after receipt of notice of such breach; or
  - (b) commits a material breach of this Agreement which is not capable of remedy;

and, for the avoidance of doubt, any breach which consists of a failure by either party to perform an obligation under this Agreement within any period required or by any date specified under this Agreement shall be deemed to be capable of remedy if such obligation is performed by such party within the 14 day cure period specified in clause 13.3(a) above.

- 13.4 The Licensors shall have the right to terminate this Agreement by notice forthwith if the Licensee:
  - (a) is dissolved (other than pursuant to a consolidation, amalgamation or merger);
  - (b) becomes insolvent or is unable to pay its debts (as that term is defined in section 123
    of the Insolvency Act 1986) or falls or admits in writing its inability generally to pay its
    debts as they become due;
  - (c) makes a general assignment, arrangement or composition with or for the benefit of its creditors;
  - (d) institutes or has instituted against it a proceeding seeking a judgment of insolvency or bankruptcy or any other relief under any bankruptcy or insolvency law or other similar law affecting creditors' rights, or a petition is presented for its winding-up or liquidation, and, in the case of any such proceeding or petition instituted or presented against it, such proceeding or petition:
    - results in a judgment of insolvency or bankruptcy or the entry of an order for relief or the making of an order for its administration, winding-up or liquidation; or

- (ii) is not dismissed, discharged, stayed or restrained in the case of a winding-up petition within 14 days or in the case of an administration petition within 2 days, of the institution or presentation thereof;
- has a resolution passed for its winding-up, official management or liquidation (other than pursuant to a consolidation, amalgamation or merger);
- seeks or becomes subject to the appointment of an administrator, provisional liquidator, conservator, receiver, trustee, custodian or other similar official for it or for all or substantially all its assets;
- (g) has a secured party take possession of all or substantially all its assets or has a distress, execution, attachment, sequestration or other legal process levied, enforced or sued on or against all or substantially all its assets and such secured party maintains possession, or any such process is not dismissed, discharged, stayed or restrained, in each case within 30 days thereafter; or
- (h) causes or is subject to any event with respect to it which, under the applicable laws of any jurisdiction, has an analogous effect to any of the events specified in clauses (a) to (g) (inclusive).
- 13.5 The licences granted under clause 2 of this Agreement are so granted on the basis of the representations made by the Licensee in part A of the application form.
- 13.6 Termination of this Agreement for whatever reason shall be without prejudice to any rights which have already accrued to the parties under this Agreement.

#### 14. Effect of Termination

- 14.1 Upon termination of this Agreement all licences granted under this Agreement shall terminate and the Licensee shall immediately cease to be licensed by the Licensors for the reproduction or communication to the public of Repertoire Works via the Licensed Services. Termination shall be without prejudice to the ability of Users to retain a copy of a Repertoire Work supplied to them under a Permanent Download Service.
- 14.2 Clauses 5, 6, 7, 8, 9, 14, 11, 17 and 18 shall survive the termination of this Agreement, but only in relation to the Licensee's activities during the Term.

# 15. No Assignment

- 15.1 Subject to clause 15.2, the licences granted under this Agreement are personal to the Licensee and the Licensee may not assign, sub-license or otherwise transfer any or all of its rights or obligations under this Agreement without the written agreement of both MCPS and PRS except where the Licensee wishes to assign or transfer its rights and obligations to a direct or indirect subsidiary or direct or indirect holding company of the Licensee in which case the Licensors may not unreasonably withhold consent.
- 15.2 Subject always to the other provisions of this Agreement, the Licensee shall be permitted to use the services of a third party in operating the Licensed Services, provided that:
  - the Licensee retains complete control and direction over the provision of the Licensed Services to Users; and
  - (b) the Licensors are able to audit such third party in accordance with clause 11.
- 15.3 The Licensee shall include the following provisions in its terms and conditions for the supply of the Licensed Services to Users:
  - (a) that any Repertoire Works communicated to Users may only be copied as permitted under this Agreement or by applicable law; and

- (b) that, save as may be permitted by law, no attempt shall be made by Users to:
  - remove or alter any rights management or identifier information that may be associated with any Repertoire Works; or
  - circumvent any technical measures associated with any Repertoire Work which are designed to prevent or restrict the unauthorised use of any Repertoire Work.
- 15.4 The Licensee shall, upon request by either Licensor, supply such Licensor forthwith with a copy of the Licensee's standard terms and conditions applying to the provision of any or all of the Licensed Services.

#### 16. Controlled Composition Agreements

Where any person, firm or company is or becomes a member of either of the Licensors or an Associated Society and that party itself or that party's predecessor in title or grantor has a current contract with the Licensee or the Licensee's predecessor in title or grantor:

- (a) to the extent that such contract would otherwise apply in relation to the grant of any or all of the licences referred to in this Agreement and/or the terms and conditions on which such licences are granted, the terms and conditions of this Agreement shall during the subsistence of this Agreement replace the terms and conditions of that contract; and
- (b) upon the written request of the Licensee the relevant Licensor will provide the Licensee with evidence that the relevant Member has become a Member and has given the relevant Licensor or the relevant Associated Society authority to bind the Member as regards this Agreement.

This clause does not prevent the Licensee from obtaining a licence only in relation to the rights referred to in clause 2.1 from the relevant party owning or controlling such rights where such licence is validly obtained whether before or after the Commencement Date but other than as a result of a contract with an individual composer or recording group.

For the avoidance of doubt, where the Licensee is exploiting the relevant rights outside the United Kingdom, and has in relation to such exploitation a valid licence from the Associated Society entitled to grant that licence in relation to the relevant territory or territories, this clause 16 does not operate so as to override the terms of that licence.

#### 17. Notices

- 17.1 Except where expressly stated otherwise, any notice or other written communication given under or in connection with this Agreement shall only be effective if it is in writing. Faxes and e-mails are permitted save that notices under clause 13 of this Agreement shall not be served by e-mail. In the absence of any legitimate electronic signature system, either party shall be permitted to require the confirmation in writing (signed by an authorised signatory) of any notice originally sent by email.
- 17.2 The address for service of any party shall be its registered office marked for the attention of the Chief Executive or Managing Director, or, if any other address or addressee for service has previously been notified to the server, to the address so notified. A single notice served on or sent to the Alliance and addressed to either Licensor shall be treated as validly served on both Licensors.
- 17.3 Any such notice or other written communication shall be deemed to have been served:
  - (a) if personally delivered, at the time of delivery;

- (b) if posted, at the expiry of two business days or in the case of airmall four business days after it was posted;
- (c) if sent by facsimile message or e-mail, at the time of receipt of transmission (if received during normal business hours that is 09.30 to 17.30 local time) in the place to which it was sent or (if not received during such normal business hours) at the beginning of the next business day at the place to which it was sent.
- 17.4 In proving service of a notice it shall be sufficient proof that personal delivery was made, or that such notice or other written communication was properly addressed stamped and posted or in the case of a facsimile message or e-mail that an activity or other report from the sender's facsimile machine or computer can be produced in respect of the notice or other written communication, in the case of a fax, showing the recipient's facsimile number and the number of pages transmitted.

## 18. Miscellaneous

- 18.1 No delay or omission in exercising any right or remedy hereunder shall operate as a waiver thereof or of any other right or remedy and no single or partial exercise thereof shall preclude any other or further exercise thereof or the exercise of any other rights or remedies. No waiver shall be binding or effectual for any purpose unless expressed in writing and signed by the party giving it and any such waiver shall be effective only in the specific instance and for the purpose given.
- 18.2 This Agreement sets forth the entire agreement of the parties in relation to the subject matter hereof and each of the parties hereto acknowledges that it has not entered into this Agreement in reliance on any representation or term not contained in this Agreement. This Agreement shall not be modified or varied except by a written instrument signed by the parties hereto.
- 18.3 The headings to the clauses in this Agreement are included for ease of reference only and are not part of this Agreement and are not to be taken into account in its construction.
- 18.4 The parties shall (and shall procure that any other necessary party within its control shall) execute and do all such documents acts and things as may be reasonably be required on or subsequent to completion of this Agreement for securing each of the obligations of the respective parties under this Agreement.
- 18.5 If this Agreement creates any rights which would in the absence of this provision be enforceable by any person not a party to this Agreement, such rights shall not be enforceable.
- 18.6 This Agreement shall be construed according to the laws of England and Wales and the parties agree to submit to the jurisdiction of the English Courts.

Signed by:	Signed by:
Duly authorised signatory on behalf of MCPS	Duly authorised signatory on behalf of PRS
Name:	Name:
Position:	Position:
	Date:

Ouly authorised signatory on behalf o Licensee	of the
Name:	
Position:	
Date:	

•

# **Definition of Music Service**

"Excluded Service"	means any service (or the relevant part of a service) which is a Podcasting Service or which falls within the scope (from time to time) of any of the following MCPS/PRS licensing schemes:
	(a) MCPS Ringtone licensing scheme;
	(b) PRS - Ringtone licensing scheme;
	(c) MCPS – Karaoke and MIDI scherne;
	(d) MCPS – Music-on-hold scheme;
	(e) MCPS – Supply of Background Music Services scheme;
	(f) MCPS – Premium Telephone Line Services scheme;
	(g) PRS – Premium Telephone Line Services scheme; or
	(h) MCPS & PRS Joint Ringback scheme; or
	(i) MCPS & PRS B2B Music Preview scheme.
"Limited Download / On- Demand Streaming Service " or "LD/ODS Service"	means a service (or the relevant part of a service), other than an Excluded Service, whereby a User may receive a Musical Work by streaming on-demand via a Network (where the time and place at which such Musical Work is received is selected by the User) and/or may download via a Network that Musical Work but where such download may not be retained by the User on a permanent basis. Any service falling primarily within the foregoing definition but which also includes elements which fall within the definition of Premium and Interactive Webcasting Service and/or Pure Webcasting Service shall be deemed in its entirety to be a LD/ODS Service.
"Music Service"	means a service or the relevant part of a service which is:
	(a) a Permanent Download Service;
	(b) an LD/ODS Service;
	(c) a Premium and Interactive Webcasting Service;
	(d) a Pure Webcasting Service; or
	(e) any combination of (a) to (d) above.
"Network"	means the internet, a mobile network or any other wired or wireless network.
"Permanent Download Service"	means a service (or the relevant part of a service), other than an Excluded Service, by which a Musical Work is communicated to the public via a Network in the form of a

	download and where such download may be retained by the User on a permanent basis.
"Podcasting Service"	means a service (or the relevant part of a service) whereby "programmes" (as opposed to individual Musical Works) may be downloaded to Users' Data Storage Devices (whether permanently or temporarily) and any tracks included within such programme cannot be disaggregated.
"Premium and Interactive Webcasting Service"	means a service (or the relevant part of a service), other than an Excluded Service, by which Musical Works are communicated to the public via a Network and:
	(a) no permanent or temporary copy of any Musical Work is retained by the User; and
	(b) such service is neither a Pure Webcasting Service, a Permanent Download Service, a LD/ODS Service or a Simulcast Service.
"Pure Webcasting Service"	means a service (or the relevant part of a service), other than an Excluded Service or a Simulcast Service, by which Musical Works are broadcast (as that term is defined in the Act as at 1 July 2006) to Users via a Network. For the avoidance of doubt, to constitute a Pure Webcasting Service, there must be:
_	(a) no interactive functionality, for example (without limitation), no use of controls that enable the User to pause, skip, move forward or backwards through the stream;
	(b) no personalisation of the service by the User or the ability for the User to offer preferences which then dictate the tracks that are provided to that User, for example (without limitation), no ability for the User to rate tracks so as to influence subsequent tracks that are played;
	(c) no advanced notification to the User of titles of specific tracks to be played or specific albums from which tracks will be played (other than the introduction of the next track in DJ led services);
	(d) in any 3 hour period:
	(i) no more than 3 songs from a particular album (including no more than 2 consecutively);
	(ii) no more than 4 songs from a particular artist or from any compilation of tracks (including no more than 3 consecutively);
	(e) no archived programmes less than 5 hours in duration or available for more than 2 weeks; and
	(f) no continuous programmes of less than 3 hours duration; and

	(g)	comme	e technologies, insofar as such technologies are ercially available and can be implemented imposing unreasonable costs, which aim to t: .
		<b>(i)</b>	a User or any other person or entity from automatically scanning the Licensee's transmissions alone or together with transmissions by other transmitting entities in order to select a particular sound recording to be transmitted to the User; and
		(ii)	a User from making copies, other than transient copies, of the sound recordings; and
	(h)	the dev	matic or intentional cause by the Licensee of rice receiving a transmission to switch from one n channel to another.
"Simulcast Service"		006) of a	adcast (as that term is defined in the Act as at 1 programme via a Network where such
	(a)	via a tra	Itaneous with the broadcast of such programme aditional terrestrial, satellite or cable television or ervice; and
	(b)		e from the website or other service of the ing broadcaster.
"Special Webcasting Service"	Webca record are by by a si	sting Se ings of th a single	Webcasting Service or Premium and Interactive rvice where more than 50% of the sound be Musical works communicated to the public (i) artist or band or (ii) comprise a live performance at or band with related performances by other ands.

# Royalty Fee

# 1. Definitions

"Applicable Revenue"	means the Gross Revenue less VAT (or other
1. T.C. 120012	equivalent sales tax, as applicable).
"Limited Subscription Service"	means an LD/ODS Service whereby the User:
	(a) pays to the Licensee a regular fee in order to access such service; and
	(b) gains access, only while such fee continues to be paid and controlled by appropriate DRM, to 20,000 different Musical Works or less.
"Minimum Royalties"	means the applicable minima set out in paragraphs 2.1(b), 2.2(b), 2.3(b) and 2.4(b) below.
"Mobile Licensed Service"	means a Licensed Service which is provided to Users and where, subject to paragraph 4.5 of schedule 3 the charge to the User for the Licensed Service includes all mobile network delivery charges applicable to the provision of such service.
	References to Mobile Permanent Download Service, Mobile LD/ODS Service, etc. shall be construed accordingly.
"PC Subscription Service"	means an LD/ODS Service whereby the User:  (a) pays to the Licensee a regular fee in order to access such service; and
	(b) gains access, only while such fee continues to be paid and controlled by appropriate DRM, to in excess of 20,000 different Musical Works; and
	(c) is able, only while such fee continues to be paid and controlled by appropriate DRM, to store and listen to Musical Works accessed through such service on a personal computer; and
	(d) is unable to store and listen to Musical Works accessed through such service on a portable Data Storage Device (such as an mp3 or WMA player).
"Portable Subscription Service"	means an LD/ODS Service whereby the User:
	(a) pays to the Licensee a regular fee in order to access such service; and
	(b) gains access, only while such fee continues to be paid and controlled by appropriate DRM, to in excess of 20,000 different Musical Works; and

	T
	(c) is able, only while such fee continues to be paid and controlled by appropriate DRM, to store and listen to Musical Works accessed through such service on a personal computer; and
	(d) is able, only while such fee continues to be paid and controlled by appropriate DRM, to store and listen to Musical Works accessed through such service on a portable Data Storage Device (such as an mp3 or WMA player).
"Subscriber"	means a User who is a subscriber to a PC Subscription Service, a Portable Subscription Service, a Limited Subscription Service or a Webcasting Subscription Service as the case may be.
"Subscription Permanent Download Service"	means a Permanent Download Service whereby the User pays to the Licensee a regular periodic fee in order to obtain permanent downloads.
"Webcasting Subscription Service"	means a Pure Webcasting Service or Premium and Interactive Webcasting Service (but not a Special Webcasting Service) whereby the User pays to the Licensee a regular fee in order to access such service.

## 2. Royalty Fee

- 2.1 In respect of a Permanent Download Service the Royalty Fee per download shall be the higher of:
  - (a) 8% of the Applicable Revenue; and
  - (b) either (subject to (c) and (d) below):
    - (i) £0.04 in respect of each Musical Work downloaded per download; or
    - (ii) if a Musical Work is downloaded as part of a bundle of eight to twelve (inclusive) Musical Works, £0.035 in respect of each Musical Work downloaded.
    - (iii) if a Musical Work is downloaded as part of a bundle of thirteen to seventeen (inclusive) Musical Works, £0.03 in respect of each Musical Work downloaded.
    - (iv) if a Musical Work is downloaded as part of a bundle of eighteen to twenty-nine (inclusive) Musical Works, £0.025 in respect of each Musical Work downloaded.
    - (v) if a Musical Work is downloaded as part of a bundle of thirty or more Musical Works, £0.02 in respect of each Musical Work downloaded.
  - (c) Where a single Musical Work is available for download and:
    - (i) the relevant sound recording of such Musical Work was originally released (whether in a physical or online form) two or more years previously; and

(ii) it is sold for a retail price of 49 pence (including VAT) or less;

then the minimum royalty applicable under paragraph 2.1(b)(i) above shall be £0.03.

- (d) Where a bundle of Musical Works is available for download and:
  - (i) every sound recording of a Musical Work in the bundle was originally released (whether in a physical or online form) two or more years previously; or
  - the same bundle was originally released (whether in a physical or online form) two or more years previously;

then the minimum royalty applicable under paragraph 2.1(b)(ii) above shall be £0.03 or, for bundles of more than twelve tracks, shall be as set out in paragraphs 2.2(b) (iii) to (v) above.

- (e) For the purposes of paragraph 2.1 (a) to (d) above, a collection of tracks shall only be considered a bundle if is (i) put together or otherwise approved by the relevant record company (or companies) or (ii) put together by the Licensee (provided permission of the owner of the relevant sound recording rights or artist (or artists) has been obtained and that the bundle is purchased as a whole. Collections of tracks assembled by Users are therefore not "bundles" for the purposes of paragraphs 2.1 (a) to (d).
- (f) Tracks which consist of either public domain Musical Works or non-music works (for example, spoken word) shall not be regarded as "tracks" for the purposes of counting the number of Musical Works in a bundle.
- (g) For the avoidance of doubt and for the purposes of paragraphs 2.1(b) (ii) to (v) above, by way of example, in respect of a bundle of 15 different sound recordings (including mixes) of the same Musical Work, the applicable Minimum Royalty for the whole of the bundle would be 45p.
- (h) Only one royalty (or one minimum, as applicable) shall be charged for so called "dual-downloads" where for a single price, the same track is delivered to a User's PC and mobile device.
- (h) The parties shall discuss in good faith Minimum Royalties in respect of long and short tracks.
- (j) The Licensee shall be entitled to produce and include as part of a Licensed Service free audio-only or audio-visual clips of a Musical Work embodied in a sound recording of up to 30 seconds duration (or for "classical" genre up to one minute's duration) without payment of any Minimum Royalty provided that such clip is used solely to promote the sale or other use of the Musical Work as part of the Licensed Service.
- 2.2 In respect of an LD/ODS Service the Royalty Fee shall be the higher of:
  - (a) 8% of the Applicable Revenue; and
  - (b) either:
    - (i) where the Music Service is a Portable Subscription Service, £0.60 per Subscriber per Month; or
    - (ii) where the Music Service is a PC Subscription Service, £0.40 per Subscriber per Month; or
    - (iii) where the Music Service is a Limited Subscription Service, £0.20 per Subscriber per Month; or

- (iv) for all other LD/ODS Services, £0.0022 per Musical Work played.
- 2.3 In respect of a Special Webcasting Service the Royalty Fee shall be the higher of:.
  - (a) 8% of the Applicable Revenue; and
  - (b) £0.0022 per Musical Work communicated to the public per User unless the Special Webcasting Service is a Subscription Service, in which case the parties shall negotiate in good faith a minimum fee per subscriber per month to apply in place of this £0.0022. In relation to the non subscriber based minimum, for the avoidance of doubt, by way of example, a webcast of one Musical Work to 1000 Users incurs a Minimum Royalty of £2.20.
- 2.4 In respect of a Premium and Interactive Webcasting Service (other than a Special Webcasting Service) it shall be the higher of:
  - (a) 6.5% of the Applicable Revenue; and
  - (b) either:
    - where the Music Service is a Webcasting Subscription Service, £0.22 per Subscriber per Month; or
    - (ii) otherwise, £0.00085 per Musical Work communicated to the public per User. For the avoidance of doubt, by way of example, a webcast of one Musical Work to 1000 Users incurs a Minimum Royalty of £0.85.
- 2.5 In respect of a Pure Webcasting Service (other than a Special Webcasting Service) it shall be the higher of:
  - (a) 6.5% of the Applicable Revenue; and
  - (b) either:
    - where the Music Service is a Webcasting Subscription Service, £0.22 per Subscriber per Month; or
    - (ii) otherwise, £0.0006 per Musical Work communicated to the public per User. For the avoidance of doubt, by way of example, a webcast of one Musical Work to 1000 Users incurs a Minimum Royalty of £0.60.
- 2.6 For the avoidance of doubt, the calculation leading to the Royalty Fee for a particular download, subscription or stream being either (a) or (b) above (in each of paragraphs 2.1 to 2.5 above, as applicable) shall be carried out individually in respect of each download, subscription or stream (as applicable) to each User and then the Royalty Fees, calculated on that basis, for all downloads, subscriptions or streams (as applicable) to all Users shall be added together in order to calculate the full Royalty Fees due. For the purposes of the foregoing calculation, where there is any Gross Revenue arising which is not attributable to a particular download, subscription or stream (for example, revenue arising from advertising or sponsorship etc), this shall be apportioned equally between all downloads, subscriptions or streams (as applicable).

#### 3. Bundling

3.1 Where a Music Service (for the purposes of this clause "Service A") is provided to Users in conjunction with a service which is not a Music Service (for the purpose of this clause "Service B"), then the price deemed to be payable by Users ("the User Price") for the Music Service (for the purpose of calculating Gross Revenue) shall be calculated in accordance with this paragraph 3.

- 3.2 Where there are separate published prices for Service A and Service B when available alone, then the User Price shall be the proportion of the bundled price represented by the relative standalone prices for Service A and Service B.
- 3.3 Where there are no separate published prices for Service A and Service B, then the User Price shall be deemed to be the proportion of the bundled price represented by the relative standalone prices for comparable UK services for Service A and Service B. If there is more than one comparable service, then the average shall be used.
- 3.4 If in the Licensors' reasonable opinion there are no comparable UK standalone services for Service A and Service B, then the applicable Minimum Royalties shall be payable.
- 3.5 For the avoidance of doubt, where the User Price is set according to either of paragraphs 3.2 and 3.3 above, the royalty payable shall be the higher of the applicable royalty rate applied to the User Price and the applicable Minimum Royalty.

#### 4. Promotional Use

- 4.1 Provided that the owner of the copyright in the sound recording has agreed to allow the Licensee to make the sound recording available for free to Users of its Music Service and that such owner is not entitled to payment of any royalty or any other consideration whether in monetary form or otherwise in relation to such use, the Licensors shall permit Musical Works incorporated in such sound recordings to be made available for free to Users and no royalty shall be due under the Agreement in relation to such use on the terms set out in this paragraph 4, but not otherwise:
- 4.2 For a paid for PC Subscription Service, Limited Subscription Service, Portable Subscription Service and Webcasting Subscription Service, the Licensee may provide such Licensed Service to each User for free for 2 weeks as a one-off introductory discount.
- 4.3 For a paid for Subscription Permanent Download Service, the Licensee may provide such Licensed Service to each User for free for 2 weeks as a one-off introductory discount provided that the User is limited during this period to accessing no more than ten (10) Permanent Downloads of Musical Works for free.
- For a Permanent Download Service (other than a Subscription Permanent Download Service) that has annual Gross Revenue exceeding £625,000 but less than £6,250,000 (excluding VAT) calculated by reference to the four preceding full quarters' reporting data, the Licensee may make available to Users at any one time up to two (2) Permanent Downloads of Musical Works for free provided that:
  - such promotional usage is solely to promote sales of paid-for Permanent Downloads
    of Musical Works by the same artist/band or songwriter as the Permanent Download
    of the Musical Work being made available for free;
  - the length of time that any one Permanent Download of a Musical Work is made available to Users for free is limited to one period of no more than thirty (30) days;
  - (c) any vouchers issued in relation to such promotional usage must be redeemed by Users within thirty (30) days of their first distribution; and
  - (d) the publisher(s) owning the rights to the Musical Work has/have granted prior written consent to such use.
- 4.5 For a paid for Permanent Download Service that has annual Gross Revenue exceeding £6,250,000 (excluding VAT) calculated by reference to the four preceding full quarters reporting data, the Licensee may make available to Users at any one time up to five (5) Permanent Downloads of Musical Works for free provided that it satisfies each of the terms set out in paragraphs 4.4 (a) to (d) above.

- 4.6 For the avoidance of doubt, where prior written consent of the publisher (s) owning the rights to the Musical Work is required pursuant to paragraphs 4.4(d) and 4.5, the Licensors shall put in place reasonable procedures to seek to ensure that such consent is granted or withheld within five (5) working days of the Licensee's written request, and if such consent is provided by the publisher(s), then the Licensors shall forthwith provide any additional consents in relation to such use of the rights which they or either of them administer.
- 4.7 For the avoidance of doubt, the Licensee may without restriction make the Licensed Services available for free over and above the limits set out in paragraphs 4.1 to 4.6 above, but the applicable Minimum Royalties shall apply in respect of any free access or downloads over and above such limits.
- 4.8 Insofar as any particular music publisher shall request that the promotional allowances set out above should be exceeded for the purposes of a specific promotion, the Licensors shall consider such request in good faith.

# 5. Provision of Music Services to Users outside the United Kingdom

Without prejudice to the territorial restrictions set out in this Agreement (and, for the avoidance of doubt, subject to the availability of the repertoire), where this Agreement is to cover the accessing by Users of the Licensed Service(s) from outside the UK, then the Licensee acknowledges that the Licensors may require that the royalties payable in respect of the Licensed Service(s) being accessed outside the UK shall be calculated on the basis of the rates at that time published as applying in the additional countries in question. If the Licensors require such rates to be payable and the Licensee does not agree to pay on the basis of those rates, then it may not be possible for this Agreement to cover the availability of the Licensed Service(s) outside the UK. The Licensors and Licensee shall discuss such issues in good faith, but if no agreement can be reached by such time as either party may decide at its discretion, then that party may terminate this Agreement (only in respect of the availability of the Licensed Services outside the UK) by giving the other party one month's written notice.

## **Definition of Gross Revenue**

#### "Gross Revenue"

means, subject to the provisions of this Schedule:

- (a) all revenue received (or receivable) by the Licensee from Users in relation to the provision of the Licensed Services; and
- (b) all revenue received (or receivable) by the Licensee as a result of the placement of advertising on or within the Licensed Services; and
- (c) all revenue received (or receivable) by the Licensee as sponsorship fees in relation to the provision of the Licensed Services; and
- (d) all revenue received (or receivable) in the form of commissions from third parties as a result of Users accessing and/or purchasing from a service of a third party via the Licensed Services; and
- (e) any other revenue received (or receivable) by the Licensee arising in relation to the provision to Users of the Licensed Services (including, without limitation, such revenue received in relation to donations, barter or contra deals, such deals to be valued for these purposes);

and in each of the above cases such revenue shall, for the avoidance of doubt, include any such revenue whether received or receivable by the Licensee or any associate, affiliate, agent or representative of such party.

Subject to the remainder of this schedule 3, there shall be no other deduction or set-off from the above revenues other than refunds to Users for services that they were unable to use due to technical faults in the Licensed Services.

#### 1. General

- (a) For the avoidance of doubt, the definition of Gross Revenue above shall be applied in conjunction with the provisions on bundling set out in paragraph 3 of schedule 2.
- (b) For the further avoidance of doubt, revenue which relates solely to services other than the Licensed Service shall not be included in the Gross Revenue. By way of example, this includes:
  - (i) revenue from non-music voice, content and text services;
  - (ii) revenue obtained solely in relation to non-music products and services supplied by the Licensee even if such product or service is obtained from

within the Licensed Service (eg. an artist related T-Shirt purchased from the Licensed Service).

(c) For the further avoidance of doubt, Gross Revenue shall not be reduced by credit card commissions or similar payment process charges.

# 2. Advertising and Sponsorship Revenue

- (a) Gross Revenue obtained in the form of advertising or sponsorship revenue shall be reduced by 5% to reflect the costs of obtaining it.
- (b) By way of clarification, advertising and sponsorship revenue shall be included in the definition of Gross Revenue if it is derived in relation to pages within the Licensed Service, including music related pages which do not contain or enable direct access to Repertoire Works (eg. content consisting of concert or music reviews).
- (c) Where advertising and sponsorship revenue is derived from pages which include both music and non-music related content or services (eg. on a home page or sub-home page of a service), then a reasonable proportion of such revenue shall be included in the Gross Revenue (such proportion to reflect the relative dominance of the music and non-music content).

#### 3. Audio-Visual

- (a) Prior to the Date of Convergence (as defined below), Gross Revenue generated in relation to the use of Repertoire Works in Audio-Visual Material shall be discounted by 15%.
- (b) If and when Convergence has been reached shall be assessed by the Licensors using reporting data submitted to it and this shall then be notified to the Licensee. If any dispute arises between the Licensors and the Licensee regarding this assessment, it shall be subject to expert determination by an independent accountant to be nominated by the President for the time being of the Institute of Chartered Accountants, in the absence of agreement between the Licensors and Licensee.
- (c) "Convergence" means the aggregate number of unit sales in the UK of Permanent Downloads in the form of Audio-Visual Material licensed under this Agreement is the same or greater than the aggregate number of unit sales in the UK of Permanent Downloads during any period of three Months prior to or ending on 30 June 2009 calculated by reference to not less than 80% of the units of each form of Permanent Download sold and the "Date of Convergence" means the last date of that three Month period. The Licensors shall use reasonable endeavours to inform the Licensee from time to time of the anticipated Date of Convergence.
- (d) For the avoidance of doubt, the discount set out in this clause 3 applies only to Gross Revenue and the Minimum Royalties shall not be reduced in relation to Repertoire Works in Audio-Visual Material licensed under this Agreement.

#### 4. Mobile Delivery

- 4.1 In relation to Mobile Licensed Services, for the purpose of calculating Gross Revenue under this Agreement, then, subject to paragraphs 4.2 to 4.5 below, the price paid by the User shall be reduced by:
  - (a) 15% for Mobile Permanent Download Services; or
  - (b) 7.5% for all other Mobile Licensed Services.

- 4.2 The reduction referred to in paragraph 4.1 (a) above will:
  - (a) no longer apply, in relation to Mobile Permanent Download Services, from the point in time when the price of such downloads converges to 117.65% or less than the weighted standard average price of downloads (of a similar type) obtained from other Permanent Download Services. This will be deemed to have taken place when the weighted average sales price of downloads (of a similar type) obtained from Mobile Permanent Download Services operated by the top 5 (by royalty value paid to the Licensors in the previous 4 Quarters) UK mobile network operators in a given Quarter is 117.65% or less than the weighted standard average price of downloads obtained from the top 5 (by royalty value paid to the Licensors in the preceding 4 Quarters) other Permanent Download Services for the same Quarter; or
  - (b) not apply, in relation to a particular download obtained from a Mobile Download Service, if the retail price of that download is 117.65% or less than the weighted standard average price of a similar type of download obtained from other Permanent Download Services (to be calculated as set out in clause 4.2(a) above).
- 4.3 The reduction referred to in paragraph 4.1(b) above will no longer apply for a particular Mobile Licensed Service where the retail price of that service is 117.65% or less than the weighted standard average price of the top 5 (by royalty value paid to the Licensors in the preceding 4 quarters) equivalent Music Services (which are not Mobile Licensed Services) in the preceding Quarter
- 4.4 For the avoidance of doubt, for the purpose of paragraphs 4.1(b) and 4.3 above, where a Mobile Licensed Service is launched at a price point that means that it does not qualify for the reduction under paragraph 4.1(b) above, if the relative price of that service and the non-mobile services referred to in paragraph 4.3 change to the extent that it would subsequently qualify for the reduction, then it shall be entitled to the reduction in paragraph 4.1(b) above until such time as the threshold in paragraph 4.3 above is reached again.
- Where content and delivery charges for music are quoted separately for a Mobile Music Service, then the royalty charged for the provision of such music shall be the equivalent amount to the royalty charged to that Licensee when the content and delivery charges for equivalent music products are quoted as a single price. Where the Licensee does not provide equivalent music products on the basis that content and delivery prices are quoted as a single price, then such provision of music shall be excluded from this Agreement and royalty terms shall be subject to a separate discussion in good faith between the parties.
- 4.6 All references in this paragraph 4 to prices shall be deemed to be excluding VAT or other equivalent sales taxes.
- 4.7 In the event of any dispute between the parties over the implementation of paragraphs 4.1 to 4.6 above, such dispute shall be subject to expert determination by an independent accountant, to be nominated by the President for the time being of the Institute of Chartered Accountants in the absence of agreement between the parties.

#### 5. Vouchers

- This paragraph 5 applies only to a Licensee which runs a voucher scheme where each issued voucher ("the Voucher"):
  - (a) is issued in conjunction with a festival or other musical event, and the voucher contains no branding other than that of the Licensee, the band/artist performing at the festival or other musical event, and/or the festival or other musical event; or
  - (b) does not refer to any specific artist(s), band(s) or Musical Work(s).

Repertoire Works provided via a Voucher scheme which is outside this paragraph 5.1 are not licensed under this Agreement and a separate licence for premium usage of this type must be sought from the Licensors.

- 5.2 The price of each Voucher, calculated in accordance with paragraph 5.3, below shall be included in Gross Revenue if and to the extent that it is redeemed by a User or Users.
- 5.3 The price of each Voucher shall be calculated as follows:
  - (a) Where the Voucher does not contain any branding or no branding other than (i) that of the Licensee for the Licensed Service, (ii) that of a third party ordinarily engaged in the sale of vouchers to Users or (iii) as permitted under paragraph 5.1(a) above, the price shall be the price charged to each User.
  - (b) Except as set out in (a) above, where the Voucher contains branding of a third party (which includes for the purposes of this paragraph 5.3(b) an entity in the same corporate group as the Licensee) and/or is distributed as part of a promotion by a third party, the price shall be deemed to be the full usual retail price charged by the Licensee for the Music Service to which the Voucher relates.

# Online Music Services

# Music Usage Information Reporting Format

[To be added]

# SCHEDULE 5

# Form of Accounting Statement

[to be added]

# SCHEDULE 6

# **Licence Details**

Licensee	
Licensee	
Licensee Registered Address	
Agreement Number	
Licensed Services	
Additions to Definitions of	
"Territory"	
Quarterly Advance (if applicable)	
appricable)	
Commencement Date	
	<u> </u>

# Schedule 2 - Letter from Universal Music Group to MCPS and PRS

Ms Arianna Vannini

Richard Constant

General Counsel, Universal Music Group International

The European Commission
Competition DG
70 Rue Joseph II
B-1000 Bruxelles
Belgium
[ ] September 2006
Dear Ms Vannini
Case COMP/C2.39237 Universal Music Group International Complaint to the European Commission concerning Online and Mobile Licensing
As you are aware, a reference was made by the British Phonographic Industry ("BPI") to the UK Copyright Tribunal concerning a dispute with MCPS and PRS over those societies' Licensing Scheme for the provision of online and mobile music services to the public for the private use.
Following resolution of this matter between the BPI and MCPS and PRS, Universal Music Group withdraws the above Complaint in respect of MCPS and PRS.
With kind regards
Yours sincerely

# Schedule 3 - Draft Press Release

# Companies agree digital royalties deal

Companies representing a substantial majority of the UK digital music market have reached a settlement relating to the royalties paid to composers, songwriters and music publishers for the digital delivery of music (such as full track downloads but not including ringtones) for online and mobile use.

The agreement amounts to a partial settlement of a case which is due to begin at the Copyright Tribunal today (Thursday September 28).

The three-year deal has been agreed between the MCPS-PRS Alliance representing songwriters, composers and music publishers, British recorded music trade association the BPI, iTunes and four UK mobile network operators, O2, Orange, T-Mobile and Vodafone. It sets a rate of 8% of Gross Revenue less VAT to be paid to composers, songwriters and publishers when their music is offered in a digital environment, with a lower rate (6.5%) for non-on-demand services. These rates are subject to minimum royalties.

All parties to the settlement welcome this agreement, which will help drive the on-going growth of the legitimate digital music market. The UK music download market is the biggest in Europe with sales of 34m units so far this year according to the Official UK Charts Company, already more than in the whole of 2005.

The Copyright Tribunal proceedings will continue in respect of issues which remain unresolved between the MCPS-PRS Alliance and the other parties as well as a single issue that remains outstanding for the mobile network operators and iTunes.

## **Dated**

September 2006

Mechanical-Copyright Protection Society Limited

**Performing Right Society Limited** 

**British Academy of Composers & Songwriters** 

The British Phonographic Industry Limited

iTunes S.a.r.l.

O2 (UK) Limited

T-Mobile International (UK) Limited

**Vodafone UK Content Services Limited** 

Orange Personal Communications Services Limited

SETTLEMENT AGREEMENT

DentonWildeSapte...

One Fleet Place London EC4M 7WS United Kingdom T +44 (0)20 7242 1212 F +44 (0)20 7246 7777 info@dentonwildesapte.com www.dentonwildesapte.com